

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Buddelmeyer, Hielke; Mourre, Gilles; Ward-Warmedinger, Melanie

Working Paper Why do Europeans work part-time? A cross-country panel analysis.

ECB Working Paper, No. 872

Provided in Cooperation with:

European Central Bank (ECB)

Suggested Citation: Buddelmeyer, Hielke; Mourre, Gilles; Ward-Warmedinger, Melanie (2008) : Why do Europeans work part-time? A cross-country panel analysis., ECB Working Paper, No. 872, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/153306

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WORKING PAPER SERIES NO 872 / FEBRUARY 2008

WHY DO EUROPEANS WORK PART-TIME? A CROSS-COUNTRY PANEL ANALYSIS

by Hielke Buddelmeyer, Gilles Mourre and Melanie Ward





NO 872 / FEBRUARY 2008

WHY DO EUROPEANS **WORK PART-TIME?**

A CROSS-COUNTRY PANEL ANALYSIS¹

Hielke Buddelmeyer², Gilles Mourre³. and Melanie Ward⁴













In 2008 all ECB publications feature a motif taken from the €10 banknote.



This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=1091120.



and Social Research, University of Melbourne, Alan Gilbert Building, 7th floor, 161 Barry Street, Carlton 3053 VIC, Australia; e-mail: hielkeb@unimelb.edu.au 3 Corresponding author: European Commission, DG Economic and Financial Affairs (ECFIN) and Free University of Brussels (ULB, SBS, CEB), Contact address: BU-1 4/168, European Commission, Rue de la Loi 200, B-1049 Brussels, Belgium; e-mail: gilles.mourre@ec.europa.eu

4 European Central Bank and IZA; Contact address: European Central Bank, Kaiserstrasse 29, 60311 Frankfurt am Main, Germany; e-mail: melanie.ward-warmedinger@ecb.int

© European Central Bank, 2008

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone +49 69 1344 0

Website http://www.ecb.europa.eu

Fax +49 69 1344 6000

All rights reserved.

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the author(s).

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb. europa.eu/pub/scientific/wps/date/html/ index.en.html

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

CONTENTS

Ab	Abstract						
No	n-tec	chnical summary	5				
1 Introduction							
2	Som	he theory: the factors influencing					
	part	-time employment	9				
	2.1	Business cycles	9				
	2.2	Labour market institutions					
2	2.3		13				
3	1 he	Framework of the empirical analysis	15				
	3.1	Estimation strategy	15				
1	J.2 The	offect of the husiness evale on part time	10				
4	emn	lowment in the short to medium run	20				
5	The	influence of institutions and other structural ables on the part-time employment rate					
	in th	ne longer run	23				
	5.1	Using a panel of OECD countries with general labour market institutions	23				
	5.2	Employing a panel of EU countries					
		with additional institutional and					
		structural variables	24				
	5.3	The relative importance of the business					
		cycle and institutional and structural	24				
(0	variables, a contribution analysis	20				
6	Con	clusion	28				
Re	feren	ices	29				
An	nex	1: Data	33				
An	Annex 2: Econometric results 37						
Eu	European Central Bank Working Paper Series 47						

Abstract

This empirical paper seeks to determine the relative contribution of the business cycle and structural factors to the development of part-time employment in the EU-15 countries over the 1980s and 1990s, exploiting a panel of EU countries. In the short-run, the business cycle is found to exert a short-term negative effect on part-time employment developments, which is consistent with firms utilising part-time work to adjust their labour force to changing economic conditions. Institutions and other structural factors such as changes in legislation affecting part-time employment are found to be key drivers of the rate of part-time employment, significant in the longer run. Overall, although the role of individual factors differs in the 1980s and 1990s, a contribution analysis considering the most significant factors shows that the main structural and institutional variables generally well explain the development in the part-time employment rate in the EU countries, which is not the case in the United States.

JEL Classification: J21, J22, J28, J68

Keywords: part-time employment, working time organisation, the business cycle, labour supply, labour market policies, institutions, regulations.

Non technical summary

A number of industrialised countries recorded a fast increase in part-time employment over the past two decades. Whereas the US experienced a slight fall in the share of part-time employment in total employment during the late 1980s and 1990s, the latter increased strongly in the EU-15, Japan and Canada. Recent papers have suggested that total employment growth (in terms of persons employed) in the 1990s benefited from the development of part-time employment in the European countries.

The promotion of part-time work may be an important measure through which the flexibility of labour markets can be increased, despite some possible drawbacks of this type of job in terms of substitution effects with full-time jobs and reduced career and wage prospects. On the labour demand side, it may allow employers to adjust hours worked to cyclical conditions more easily, facilitating adjustment of production and labour costs. On the labour supply side, part-time work may increase the labour market choices open to individuals and may thus improve social welfare.

Against this backdrop, this paper has three purposes. First, it reviews the determinants of parttime employment, as identified by the economic literature. Second, it aims to determine the role of the business cycle and structural factors in the development of part-time employment, using a panel of EU-15 countries. This could incidentally cast some light on whether part-time jobs have been used as a flexible work arrangement by firms in the EU-15, as has been shown for some specific countries and sectors. Third, the main objective of the paper is to measure the *relative importance* of each determinant of part-time employment over time. This justifies the macro perspective of the paper and its comprehensive approach (encompassing a large set of determinants), since the relative contribution of the relevant macroeconomic, policy and institutional factors could not have been obtained with a micro analysis or with a study focusing on one specific factor of part-time employment.

The business cycle, as measured by both the output gap and real GDP growth, is found to exert a negative effect on part-time employment developments. This countercyclical effect is consistent with firms utilising part-time employment as a way to adjust total hours worked to changing economic conditions. The "flexibility" hypothesis suggests that employers reduce the number of hours worked by current staff and/or hire new workers in part-time jobs during economic downturns, while net employment creation in upswings mostly consists of full-time jobs in order to meet the increased demand for labour resources by firms. Consistent with this, involuntary part-time employment turns out to be cyclical, being higher in troughs of economic activity. Splitting our sample by age and gender groups reveals a very significant effect of the cycle on young and male prime-age workers.

Looking at institutions, changes in legislation favourable to part-time employment are found to be effective, having a strong impact on actual part-time employment developments. Moreover, while overall employment protection legislation has no clear-cut effect, employment protection legislation for *permanent contracts* is found to be positively related to the part-time employment rate, which is consistent with the use of part-time work as a tool for enhancing flexibility in the case of rigid labour markets.

As for other structural variables, the fertility rate is found to be strongly and positively correlated with the part-time employment rate. This suggests that part-time work creates an opportunity for women to combine taking care of their children with market work. In line with conventional stylised facts, the share of the services sector in the economy and the proportion of young people in tertiary education are also significant predictors of the part time employment rate. In addition, cross-sectional analysis underscores the positive impact of the differential in hourly wages between part-timers and full-timers on the part-time employment rate. This suggests that hiring through part-time relative to full-time positions would be fairly sensitive to relative wages. This could also reflect part-time work being used by firms as a cheap and "precarious" form of employment, in line with the "insider/outsider" theory.

Finally, a contribution analysis, comparing the relative importance of each factor, finds that the main institutional and structural variables generally explain the development in the part-time employment rate in the EU countries fairly well over the full period 1983-1998, while this is not the case in the United States. Moreover, in some EU countries such as United Kingdom, Ireland, the Netherlands and Portugal, other factors, not captured by the analysis, have also contributed to increasing the part-time employment rate. While rising female participation appears to be the main driving force behind the development in part-time employment, declining fertility rates exert a dampening effect in some countries. Looking at labour market institutions, the relaxation of the legislation on part-time jobs in Belgium, Spain and France played the major role. Other institutions may also have played some part in Spain and Portugal (easing of EPL for permanent jobs) and in the Netherlands (the cut in child benefits).

The role of the various determinants of part-time work differs across countries in the 1980s and 1990s. The effect of the fertility rate was stronger in the 1980s. Child benefits had an unambiguously positive contribution in 1980s, while their contribution was weaker or negative in the 1990s. Conversely, the change in part-time employment laws and in EPL for regular contracts only played a strong role in the 1990s. Lastly, the effect of the business cycle contributes, albeit to a limited extent, to the generally stronger developments in part-time employment seen in the 1990s, compared with the 1980s. However, its role was quite negligible over the full period, as cyclical peaks and troughs offset each other.

1. Introduction

A growing part-time employment share was a main feature of a number of industrialised countries over the past two decades. While the US experienced a slight fall in the share of part-time employment in total employment during the late 1980s and 1990s, this share increased strongly in the EU-15¹, Japan and Canada in the same period. From a macro-economic standpoint, the increase in the share of part-time work may have a dampening effect on aggregate wage growth² and some recent papers have shown that total employment growth (in terms of persons employed) in the 1990s benefited from the development of part-time employment in the European countries (Garibaldi and Mauro 2002, Mourre 2006).

Furthermore, from an economic policy perspective, the promotion of part-time work may be an important measure through which the flexibility of labour markets can be increased. On the labour demand side, it may allow employers to adjust hours worked to cyclical conditions more easily, facilitating adjustment of production and labour costs. On the labour supply side, part-time work may increase the labour market choices open to individuals and may then raise their welfare.

At the same time, part-time work may have its drawbacks. Part-time workers may be subject to lower wages, be less likely to receive fringe benefits and face reduced chances for promotion (for a discussion, see e.g. OECD 1999). For employers, part-time work may be connected with higher fixed costs, such as recruitment and training costs. Furthermore, part-time jobs may crowd out full-time positions.

Broadly speaking, the literature on part-time job developments to date can be divided into two main vintages. A first branch relies on micro-econometric analysis based on firm surveys. The purpose of this work is to identify the microeconomic and individual determinants of part-time work (Haskel 1997; Houseman 2001), to illustrate the role of part-time work as a flexible work arrangement (Haskel 1997, Farber 1999, Barrett and Doiron 2001), or to explain the transition between part-time jobs and other labour market states (Blank 1994, O'Reilly and Bothfeld 2002; Buddelmeyer, Mourre and Ward 2005a). The second vintage deals with the comprehensive description of part-time employment developments based on macro-data and includes sociological and institutional qualitative evidence (Smith et al.1998; Delsen 1998; OECD 1999, Walwei 1998). In this vein, some papers focus on international comparison (Lemaitre et al. 1997; Buddelmeyer, Mourre and Ward 2005b) and attempt to understand the reason why part-time

¹ i.e. the European Union before its enlargement in May 2004, comprising 15 member states.

² This is because, first, the increase in the part-time employment rate mechanically decreases the average hours worked per individual, reducing total average compensation per employee, ceteris paribus. This is a pure measurement effect. Second, some studies have shown that the hourly wage rate received by part-time workers is lower than that of full-time workers (OECD 1999). Both of these two dampening effects on aggregate wage growth may increase with a rise in part-time employment.

work is so low in some countries such as Spain and Portugal (Ruivo et al. 1998) or so different across countries (Pfau-Effinger 1998).

The present paper belongs to the second vintage of the literature, whilst extending it through the use of a country panel of time-varying institutions and structural variables covering most EU-15 Member States. This latter methodological approach can be associated with another branch of the labour market literature (initiated by Scarpetta 1996 and extended by Belot and van Ours 2000, Blanchard and Wolfers 2000 and Nickell et al. 2003) who uses cross-sectional or pooled time-series data on indicators of labour market performance and labour market institutions to account for unemployment differentials across countries. This paper therefore fills a gap in the literature by analysing the cause of part-time job developments systematically *in a Europe-wide perspective*, including the *ex post* impact of changes in legislation regarding part-time employment³. This paper also considers the effect of the business cycle on part-time employment. Although the counter-cyclicality of the relative growth of part-time employment compared to full-time employment has often been mentioned in the literature (Delsen 1998, Lester 1999) and tackled indirectly through the cyclicality of hours worked (Millard 1999), no systematic econometric investigation has been carried out using macroeconomic data.

Thus the purpose of this paper is threefold. First, it reviews the determinants of part-time employment as identified by the economic literature. Second, it aims to identify the role of the business cycle and structural factors in the development of part-time employment in the EU-15 countries by exploiting both cross-sectional and time series variation. This may enable us to comment upon whether part-time jobs have been used as a flexible work arrangement by firms in the EU-15, as has been shown for some specific countries and sectors within the micro-econometric literature. Third, the main goal of the paper is to measure the *relative weight* of each determinant of part-time employment over time. This justifies the macro perspective of the paper and its comprehensive approach (encompassing a large set of determinants), since the relative contribution of the relevant macroeconomic, policy and institutional factors could not have been obtained with a micro analysis or with a study focusing on one specific factor of part-time employment.

The paper is organised as follows. Section 2 surveys the determinants of part-time work offered by theory. Section 3 introduces the simple econometric setting to be utilised in the remainder of the paper and describes our data sources. Section 4 analyses the impact of the business cycle on part-time employment developments and on involuntary part-time work. Section 5 considers the role of part-time legislation, labour market institutions and other structural features (of demographic, sociological or economic nature) on the rate of part-time work within EU-15 countries. Section 6 concludes.

³ It should be noted that the vast economic literature on European employment laws tends to focus primarily upon the distinction between permanent vs. short-time contracts, rather than full-time and part-time contracts.

2. Some theory: the factors influencing part-time employment

This sub-section reviews the various determinants of part-time work offered by theory which will be consequently important for our econometric analysis in the following sections. The determinants can be grouped under the headings of (i) The business cycle, (ii) Labour market institutions and policies and (iii) Other structural factors. While the economic literature generally suggests that the business cycle plays some role in the short to medium run, it also stresses the importance of various institutions and policies as well as other structural factors of sociological, demographic or economic nature in the longer run.

2.1 Business cycles

Business cycles affect movements in the part-time employment rate⁴ in the short to medium term. This implies that the rate of part-time work may respond differently to business cycle effects relative to full-time employment. The literature has identified a number of ways in which the business cycle affects the share of part-time employment (see OECD, 1995, Delsen, 1998).

On the demand side, two effects may be at play. First, there is a compositional effect. As noted by Lester (1999), shifts *between* sectors arise because sectors with a higher share of full-time employees, such as manufacturing and construction, are generally affected earlier and more strongly by the cycle than other sectors. Therefore, part-time employment may be less responsive to business cycle effects than total employment, and may show a counter-cyclical pattern.

Second, shifts *within* sectors may boost part-time employment during downturns as employers offer part-time employment as a way to adjust hours worked over the cycle, while continuing to hoard labour (Delsen 1998). This enables employees in place to avoid dismissal and job seekers to avoid long-term unemployment. Thus, during a downturn, employers may reduce the number of hours worked by their current workforce or hire new workers in part-time jobs. However, as suggested by Lester (1999) for Australia, and Farber (1999) for the US, it seems that a non-negligible proportion of full-timers moving to part-time work do not stay in their current firm, going though a period of unemployment⁵. Moreover, as some employers use part-time jobs to screen workers for full-time positions (Houseman 2001) and firms may be more demanding and more risk-averse before hiring new staff in bad times, they may prefer to systematically recruit part-timers rather than full-timers in periods of economic slowdown. In an upswing, employers may offer full-time contracts to their part-time workers (so called "tap effects") in order to increase their total stock of labour resources, as this was the case in Sweden in the 1980s (Sundström 1991). We will label the use of part-time work by employers as a means to regulate

⁴ The percentage of those working part-time relative to total employment

⁵ Farber (1999) for the US found that job losers are significantly more likely than non losers to get an involuntary part-time job, which may suggest that workers, formally in full-time jobs, go through a (short) unemployment spell before occupying a part-time job.

the size of their workforce in response to the economic cycle as the "flexibility effect". We would therefore expect the flexibility effect to exert a countercyclical impact on the rate of part-time employment⁶.

On the supply side, in an environment of falling economic activity and/or rising unemployment, workers may be more willing to consider part-time work as an alternative to their preferred labour supply choice of full-time employment. This third effect can also be viewed as a consequence of the "flexibility effect", as it also reflects the mismatch between job seekers, preferably looking for a full-time job, and firms mainly offering a part-time job. Conversely, in a period of expanding output and/or falling unemployment, workers with a preference for full-time work may be less willing to work part-time. This "involuntary part-time work effect" is expected to be negatively related to the part-time rate.

An additional labour supply effect is the "added worker effect". Here, the unemployment of a household member leads to an increase in the labour supply of other household members (Mincer 1962, Layard et *al.* 1980 and Maloney 1991). Typically, a recession causes job losses in single breadwinner household, urging the other partner (often female) to work part-time so as to limit the income loss. The empirical evidence to date however suggests that this effect is small.

A fifth effect, also of labour supply, can be identified, known as the "discouraged worker effect" (Long, 1953). A lasting period of unemployment reduces the probability of finding a job and may induce groups of disadvantaged groups not to enter in the labour market, awaiting better opportunities, given that searching for a job is a costly activity. A large set of empirical studies reports evidence supporting the discouraged worker hypothesis (Clark and Summers, 1982; Blundell, Ham and Meghir, 1998; Darby, Hart and Vecchi, 2001). During a period of economic contraction or slowdown, labour supply from low-skilled or initially inactive females, who are more likely to look for part-time jobs, may be reduced, given the lesser prospect of finding a job. Thus, the discouraged worker effect is likely to be negatively related to the part time rate.

⁶ In other words, the flexibility effect reduces the negative cyclical responsiveness of the number of those working part-time compared to those working full-time, thereby causing a countercyclical effect on the share of part-time employment in total employment.

Table 2.1
Five effects of business cycle on part-time employment rate

Labour demand
(-) compositional effect (concentration in less cyclical sectors)
(-) flexibility effect (adjustment of hours worked along the cycle)
Labour supply
(-) involuntary part time effect (enhanced willingness to work part-time to avoid unemployment)
(-) added worker effect (unemployment of a household member causing an increase in labour supply of another household member)

(+) discouraged worker effect (withdrawal from the labour market of part-time job seekers.)

===> Overall effect uncertain, but likely negative correlation between output gap and part-time increase

Reading: (-) means countercyclical effects. (+) means procyclical effects.

Thus, the effect of the cycle on the rate of part-time work will be positive or negative depending on the relative importance of these five effects (see Table 2.1). The finding of a significant counter-cyclical effect could suggest that (i) labour demand is one of the main driving forces behind developments in part-time employment and that (ii) employers use these types of jobs to adjust to the general economic situation. Indeed, the labour supply effects going in the same direction are supposed to be weak ("added worker effect") or the mirror of cyclical labour demand conditions ("involuntary part-time effect"). In contrast, the finding of no or weak cyclical patterns of part-time work relative to total employment may indicate that developments in part-time jobs rather reflect structural factors, such as the design of labour market institutions.

2.2 Labour market institutions

While the business cycle affects the short-to-medium term, developments in part-time employment, institutional and sociological factors are likely to influence the longer-term level of the part-time rate. Smith et al. (1998) survey the factors affecting the growth of part-time work in Europe. Some of these factors are not mutually exclusive and might reinforce each other.

Specific regulations on part-time work (statutory or collectively bargained) can affect part-time employment developments through three mechanisms, as described by Smith et al. (1998). First, some regulations directly affect the working-time system by restricting the use of part-time work. Second, some regulations indirectly influence part-time employment through wage regulation, the social protection system or the tax and benefit system. For example, in some countries (e.g. Portugal), hourly wages for part-time and full-time work are restricted by law to be equal (pro rata clause). Such legislation may change the structure of non-wage costs for employers, such as hours or earnings thresholds for social insurance contributions or special

subsidies or tax rebates for firms hiring part-time staff⁷. Lower pension entitlements can also restrict the labour supply of part-time jobs (Ginn and Arber 1998). A third type of regulations facilitates the conversion of full-time jobs into voluntary part-time employment in order to reconcile personal and professional lives (Genre et al. 2003).

Employment protection legislation (EPL) has become a more and more important reason for hiring part-time employees. In the late 1990s, in a context of increasing competition and uncertainty, firms realised the advantages of "atypical jobs", which were considered to be more flexible and less expensive. Part-time employment was quickly acknowledged as offering such a job type, together with temporary jobs, help agency employment, shift work, on-call jobs and atypical working hours (nights and weekends). Bentolila and Dolado (1994) call this new search for flexible staffing arrangements the "new reasons" for hiring part-time workers. The stringency of job protection legislation may have two conflicting effects. First, it can encourage the use of part-time work so as to *circumvent* highly rigid employment legislation affecting full-time jobs. Second, employment protection legislation in some countries can affect directly and adversely part-time employment by strictly limiting its use. In any case, as part-time employment in the EU-15 mostly corresponds to permanent contracts (above 80% in the late 1990s), *EPL for regular contracts* seems more relevant for understanding the level of part-time work over time and across countries. However, the small proportion of fixed-term part-time employment can be affected by *EPL for temporary contracts*⁸.

The effect of *child benefits* on part-time work is ambiguous. In general, the provision of child benefits can create an "unemployment trap", if benefits are only granted below a certain level of income⁹. Moreover, some child benefits grant additional income to parents who renounce to work in order to take care of their young children. These can have a strong adverse effect on female participation (Laroque and Salanié 2003). On the other hand, the lack of childcare benefits or a subsidised *childcare system* is a major disincentive to taking up employment. Therefore, the overall effect may depend on the shape and the composition of child benefit systems.

Unemployment benefit systems (together with other benefits) may create an "unemployment trap" through high net replacement rates and long benefit duration. As stated by Doudeijns

⁷ For instance, in 1993-2000, a specific social security contribution rebate was granted to firms in France for each recruitment in part-time, in addition to some additional social security contribution cut for the recruitment of low-paid part-timers. This coincided with a boom in part-time employment over the same period. As reported by Ruivo et al. (1998), the legislation on part-time work underwent important changes in Spain in 1994 (relaxation of the maximum daily and weekly working hours, reduction in social security contribution). This resulted in a strong increase in the part-time employment rate between 1994-1995.

⁸ A low level of EPL for temporary contacts may also make the use of part-time contracts less necessary as temporary employment can meet the need for more flexible staffing arrangements.

⁹ If child benefits are means-tested, this increases the marginal rate of taxation around the income threshold above which the incumbents are not eligible to these benefits. This might deter one member of the couple to take up a job because this may mean the loss of child benefits. However, if the eligibility threshold is high, this could encourage taking a part-time job rather than a full-time job.

(1998), this financial disincentive to take up a job has a significant impact on the decision to work part-time, since part-time jobs are not likely to yield high earnings. Conversely, "in-work benefits", the tightening of eligibility criteria and enhanced controls of their implementation may help to alleviate such a disincentive.

The tax system and in particular *high income tax* rates may deter part-time work. Where the income tax is computed on the basis of a couple's earnings rather than on an individual basis, the second earner in the couple may be taxed at a relatively high marginal rate, creating an "unemployment trap". Moreover, the existence of a dependent partner's allowance may create disincentives for the second earner to take up a job, especially low-paid part-time jobs (Jaumotte, 2003).

Trade unions have generally fought against part-time work – seen as a threat to full-time standards. Houseman (2001) found some econometric evidence of a negative relationship between trade union density and part-time employment for the US, although reverse causation cannot be ruled out^{10} . This can also explain the low part-time rate in highly unionised sectors such as manufacturing, while the services sector, less unionised, is more part-time intensive. However, high unemployment has started to change trade unions' attitudes. In this context, a high level of co-ordination between unions may facilitate the conclusion of tripartite agreements involving the government, which aim at reducing unemployment via some "work sharing" or improving working conditions by encouraging voluntary part-time employment (Genre et al. 2003).

The share of temporary jobs (short-term contracts, help agency employment) could exert a crowding-out effect on part-time employment. In Portugal and Spain (until 1994), the favourable legislation on fixed-term contracts and the sharp development in this type of job have lead to a weak development in part-time employment. Nonetheless, in many cases, the development in part-time jobs goes hand in hand with that of fixed-term contracts as complementary flexible staffing arrangements¹¹.

2.3 Other structural variables

The rise in *female participation* has occurred hand in hand with an increase in the part-time rate in many countries. While the causality is not clear between the decision to participate into the labour market and the choice of working part-time, the rising proportion of women willing to join the labour markets, mostly explained by rising levels of education, contributed to the development of part-time employment. Sociological and cultural reasons, such as the separation

¹⁰ Firms or sectors with a high proportion of part-timers are generally considered to be harder to unionise.

¹¹ In France, part-time jobs increased strongly in the service sector in the 1980s and 1990s, while temporary jobs and particularly agency employment flourished in manufacturing.

of tasks within the household and the family model, combined with institutional reasons (e.g. the lack of childcare facilities) explain in part why women are more inclined to work part-time than men. For instance, a "male breadwinner" model of family encourages women to work part-time rather than full-time (Fagan and O'Reilly 1998). Likewise, an increase in the *fertility rate* and the number of children per family may increase the need for working part-time instead of working full-time so as to reconcile professional and family life, especially when the child care system is insufficient. However, in some countries, such as Finland, the male breadwinner model was never dominant and the rise in female participation mainly induces the increase in full-time employment (Pfau-Effinger 1998). However, these variables may suffer from endogeneity problems.

The share of *employment in services* is also an important determinant of part time work in the literature to date. Traditionally, employers hire part-time workers in order to deal with regular and predictable peaks of demand ("rush hours") or to extend opening hours on evenings or weekends, beyond what the hours worked by a full-timer in a single shift would allow¹². This situation particularly concerns some service sectors, where part-time work is concentrated (retail trade, hotels and restaurant, personal services). Conversely, mass-production and capital intensive industries rely far more on full-time workers. Part-time work may also be used to substitute for absent employees (on maternity or sick leave) or to meet unexpected orders. Part-time staff therefore acts as a back-up for full-time employees (Delsen 1995). In addition, as mentioned by Delsen (1998), the issue of "gender flexibility" plays a role. Many services rely on predominantly female employees, who are more prone to accept working part-time, while maledominated sectors (industry, construction) have been more reluctant.

The schooling rate of the 15-24 aged population is likely to positively influence the part-time employment rate, as the latter is currently highest for those aged 15-24 within the EU. From the labour demand side, students offer relatively cheap and flexible labour¹³. From the labour-supply side, students are mainly looking for part-time jobs to finance their studies. Moreover, some newly graduated students may be ready to accept working part-time in bad times in order to build up professional experience and avoid unemployment spells, which might be stigmatising.

The *differential in wages and non-wage costs* between part-time and full-time employment may make part-time work advantageous to firms. OECD (1999) showed that both the hourly wage rate and the benefits received by part-time workers were lower than that of full-time workers in most OECD countries¹⁴. This remains true even after controlling for occupational characteristics of part- and full time jobs. The gap between part-time and full-time earnings is 10% on average.

¹² Therefore, employers avoid underemployment of staff in off-peak times and overtime payment in busy periods.

¹³ Young workers may be considered as "outsiders", allowing firms to make them bear a higher proportion of adjustment costs in terms of wages and working conditions. Youth labour may often consider their part-time job as temporary, limiting the risk to the firm of recruiting an inefficient worker.

¹⁴ See OECD employment outlook, June 1999.

One reason for this may be that part-time workers tend to receive less job-related training than full-timers. As suggested by Houseman (2001), a second reason might be that part-timers belong to a "secondary" labour market, and might not have as strong a bargaining position as full-time workers. Likewise, part-timers are less likely to benefit from an "efficient wage" policy, given their lower contribution in terms of number of hours to the firm¹⁵.

3. The framework of the empirical analysis

This section explains the estimation strategy and the data used to show empirically how important the factors mentioned in the previous section are in the determination of the rate of part-time work in EU-15 countries.

3.1 Estimation strategy

The part time employment rate follows a clear positive trend over time in Europe and the presence of a unit root is confirmed in most countries by running the Augmented Dickey-Fuller test and Phillips-Perron test. We first use the annual change in the part-time employment rate (ΔPT) as the dependent variable in order to address the issue of non-stationarity. As this variable offers a limited time series dimension by country (i.e. 18 time observations at most), we pool annual data across EU-15 countries. Therefore, for our analysis of the effect of the business cycle on the part-time employment rate we first use the following simple fixed-effect specification (*a*) where OG denotes a measure of the cyclical position of the economy and α_i are *i* country dummies¹⁶:

(a)
$$\Delta PT_{it} = \alpha_i + \beta OG_{it} + \varepsilon_{it}$$

Under this model (*a*), the business cycle OG is proxied by (i) estimates of the output gap and (ii) real GDP growth. The former is preferred since it offers a better measure of the cyclical fluctuation of economic activity, whilst real GDP growth also encases variation in potential output growth across countries. The output gap is defined as the difference between actual GDP and potential GDP as a percentage of potential GDP. The latter, also called potential output and calculated using the production function method, corresponds to the productive capacity of the economy, that is, the level of sustainable output, compatible with a stable inflation rate. However, since output gaps are derived from model-based estimates of potential output and are

¹⁵ More tentatively, another reason for an increase in part-time employment might be the demographic ageing, as the elderly may need more care, often provided by part-timers, and young generations may find part-time jobs more appealing. However, given the slow nature of demographic changes, these effects, playing in different directions, are likely to be of second order. Likewise, the substantial immigration observed in some EU countries might influence part-time employment, although the theoretical bond is unclear.

¹⁶ In specification (*a*), we do not use time dummies because they would bias the estimation of the effect of the economic cycle. Indeed, they would capture the component of the business cycles that is common across countries and the coefficient of the output gap would only take into account the country-specific component instead of the full cyclical effect.

consequently "non observable" variables, real GDP growth is used to check the robustness of the results.

In this model, we include lagged independent variables. According to a Wald test, two periods lags are found to be sufficient for our model, as the removal of further lags (3 and 4) does not affect the results significantly. Since the effect of the business cycle on the part-time rate may be asymmetric, a specification (iii) regresses the output gap (with lags) interacted with a dummy variable, which indicates whether the output gap is positive or negative. Finally, one may conjecture that the "flexibility effect" mentioned earlier would be more acute for countries experiencing strong structural rigidities. In these countries, which are likely characterised by a high unemployment rate, firms would use part-time work as a means of circumventing structural rigidities and adapting resources according to fluctuations in economic activity. Moreover, in the case of very high unemployment, workers may be more willing to accept a part-time job, enhancing the involuntary part-time work effect. Therefore, specification (iv) uses the output gap interacted with a dummy reflecting a high unemployment country to investigate whether the countries hit by high unemployment witness a more pronounced cyclical pattern.

Model *(a)* is straightforward, breaking down the PT rate into a cyclical component and a linear country-specific trend¹⁷. It easily resolves the problems arising from the use of non stationary variables. However, the results are constrained by the absence of quarterly data for part-time employment, which would have been more suitable to investigate its cyclical behaviour. Annual data are by definition less frequent, tends to smooth out quarterly pattern and loose information about the start of upturns and downturns.

For sake of robustness, we also complement this model in first differences with a more comprehensive model in levels. The structural changes brought out by reforms in the regulatory or legal framework of part-time employment may cause acceleration in part-time developments, which is not captured by the linear trend of model (a). This may disturb somewhat the empirical estimation of the cyclical component of part-time employment. Thus, our model (b) considers the *level* of the part-time employment rate across countries and over time as well as the impact of structural factors and institutions. It also allows us to check the robustness of the cyclical effect and control for possible omitted variable bias by taking into account additional explanatory variables. It is expressed as follows:

(b)
$$PT_{it} = \alpha_i + \alpha_t + \beta OG_{it} + \sum_{k=1}^{K} a_k X_{k it} + \sum_{j=1}^{J} b_j X_{j it} + \varepsilon_{it}$$

where α_i are *i* country dummies and α_t are *t* time dummies. X_k are *k* variables representing institutions and X_j are *j* variables capturing structural factors (of societal, demographic or economic nature).

¹⁷ The linear country-specific trends correspond to country fixed effects (i.e. country-specific intercept), as the model (a) is written in first differences.

This specification is close to that used by Nickell et al. (2001) to explain OECD unemployment, but the variables for short-term macroeconomic shocks (labour demand shock, real import price shock, real interest rate, etc.) are replaced by the output gap. Moreover, the lagged dependent variable is dropped as its presence introduces correlation between the errors and the regressors, which can bias the GLS estimate of parameters¹⁸.

As most variables are not stationary, it is crucial to check that they are linked together in the long run through a cointegrating relationship and thus that the residuals of the equation are stationary. In order to test for unit roots in the residuals we use the Fisher-Maddala-Wu test, which is suggested by Maddala and Wu (1999) on the basis of previous work by Fisher (1932). This non parametric test is used here since it does not require a balanced panel. It is also easy to run as it andcombines the observed significance levels of any type of unit-root test performed individually on N countries. If Pi denotes the *p*-value from the unit test on the i^{th} country time series, then $-2\Sigma \ln Pi$ follows a χ^2 distribution with 2N degrees of freedom. However, this test assumes the absence of cross-sectional correlation, i.e. that the observations are generated independently across countries. When this assumption is violated, which is often the case in practice, the standard test may suffer from large size distortion, as evidenced by Maddala and Wu (1999) who have computed the exact distribution of the test and corrected the critical value by bootstrapping. Here, following Nunziata (2005), we control for cross-country correlation using time dummies (which capture cross-sectional dependency caused by common timespecific components) and then we utilise the standard distribution of the test. Under the Fisher-Maddala-Wu test and using Phillips-Perron tests for individual countries, the null of no cointegration (non-stationarity of the residual) is strongly rejected for all equations in Table 5.1¹⁹. For instance, using the equation 1 of Table 5.1, the test statistic is $\chi^2(40) = 79.2$ with a pvalue of 0.02%. The test clearly rejects the hypothesis of the residuals having a unit root and points to the existence of a cointegrating relationship between the regressors 20 .

As some heteroskedasticity and first order autocorrelation is detected in both models a and b, all equations are estimated by feasible generalised least squares (see also Nickell *et al.*, 2003 and Nunziata, 2005)²¹.

To tackle possible endogeneity problems, we also use an instrumental variable method, namely the two-stage feasible generalised least squares. Although the causality may mainly run from GDP/output gaps to part-time employment, the growth rate of fixed capital investments (along

¹⁸ Given the fairly short time series dimension of the data, this bias may be not negligible.

¹⁹ The results are confirmed when using Dickey Fuller test as country-by-country unit-root test.

²⁰ Under the Maddala and Wu test and using Phillips-Perron tests for individual countries, the null of no cointegation (nonstationarity of the residual) is strongly rejected for all equations in Table 5.1. For instance, using the equation 1 of Table 5.1, the test statistic is $\chi^2(40) = 79.2$ with a p-value of 0.02%. This test relies on no cross-country correlation. The introduction of time dummies in specification (*b*) is likely to capture much of the cross-correlation in the data.

²¹ For instance, on the basis of regression 1 in Table 4.1, the hypothesis of homoskedastic residuals is strongly rejected by a LR test ($\chi^2(13)=72.8$), while the hypothesis of non-serially correlated residuals is clearly rejected by a LM test ($\chi^2(1)=94.2$).

with its lagged values) and the growth rate of trade and services exports are used as instruments in specification (*a*). Indeed, whereas employment developments directly influence output growth through household disposable income and household consumption, trade and services exports and fixed capital investments are not driven by national employment, at least in the short run. Instead they are strongly correlated with output and table 4.2 shows that the two variables are important explanatory factors of the output gap²². This suggests that the two-stage least squares estimate should not suffer from the weak instrument problem.

In order to further gauge the robustness of the results to the possible endogeneity of regressors, we also run granger-causality type of equations (a'), where the business cycle variable is lagged by one year. The estimate (a') confirms the significance and the sign of the output gap coefficient, although its magnitude is lower than in estimates (a) (see Table 4.3).

(a')
$$\Delta PT_{it} = \alpha_i + \beta OG_{it-1} + \varepsilon_{it}$$

In specification (*b*), the issue of endogeneity is very likely to particularly affect the relationship between the part-time employment rate and participation. Moreover, a growing share of part-time jobs might make policy makers more responsive to part-time worker preferences. In this respect, the European Union directive on part-time work could be seen as a European Union wide response to the growing importance of part-time workings.²³ Therefore, we instrument the structural and institutional variables by their values lagged by two years. One year lag does not suffice because of the presence of first order autocorrelation in the residuals.

3.2 Data

The data used in this paper are presented more in detail in Annex 1. The part-time employment rate (PTR) is defined as the percentage of workers working part-time relative to total employment. The part-time employment rate, as well as the share of involuntary part-time employment, is derived for each country from the annual European Community Labour Force Survey, which has been conducted every year in the spring since 1983 and compiled by Eurostat. The definition of "part-time" is based on individuals' declarations of working status (part-timer versus full-timer), rather than on a precise measure of hours worked.

²² The instruments are selected from equation 1, which contains the largest number of highly significant variables.

²³ Possible endogeneity with other policy variables is less obvious. Indeed it is reasonable to assume that the developments in labour market institutions and policies in EU countries have been influenced by the general situation of the labour market in each country (as reflected by the unemployment rate), the causation from part-time employment rate to institutions seems to be fairly unlikely. Moreover, as pointed out by Nunziata (2001), the reasons behind institutional and policy changes are far from evident. The pattern of political decision making in each country would tend to indicate that other important factors such as the timing of election, the policy makers' agenda and the average voters' preference may have played a crucial role in shaping the institutional and policy changes.

Data on business cycle (output gap, GDP growth) comes from OECD Economic Outlook. The output gap is measured as the percentage difference between actual GDP in constant prices and potential GDP estimated by the OECD²⁴.

Data on labour market institutions and structural variables are taken from various sources, in particular the recent macro-economic literature on labour markets. Most of these data on national labour market institutions were available only until 1997-1998 when the study was carried out. It should be noted that the empirical models are estimated with annual data, which may underestimate the full effect of the business cycle by not taking into account intra-annual cyclical variations. Indeed, the annual indicator of output tends to smooth out the real cyclical fluctuations. The choice of annual data relates to statistical limitations. While cyclical indicators can be available on a quarterly basis, part-time employment series are only available on an annual basis as well as data on labour market institutions and structural variables. The use of panel data approach however increases the number of observations and circumvents this lack of time length²⁵.

The series capturing the effect of part-time regulation on the part-time rate deserve particular attention. This is a vector of country-specific trends which start from the year in which regulations of part-time employment were relaxed in each country²⁶. The motivation for the creation of this variable is that we wish to capture the additional annual increase in the part-time employment rate that follows the change in part-time regulation²⁷. The latter is identified using the information provided by EIRO and OECD (1999). The starting date of each trend therefore varies by country. For the countries where part-time regulation has not changed (Ireland, Portugal, Sweden and the United Kingdom), the value of the variable is set at zero for the complete period 1983 to 2001. To avoid an "outlier" bias, we have also constructed the "policy"

²⁴ For further details, see Giorno *et al.*, "Potential output gaps and structural budget balances", OECD Economic Studies, N.24, 1995/I.

²⁵ A panel data analysis appears adequate. A cross-section study only exploiting the time average of each variable would face the problem of the relatively limited cross-sectional dimension (21 OECD countries and 14 EU countries) and will lose some time-series information. Table 3.1 shows that the time variation ("within") is relatively important for some structural variables (female participation, fertility rate, schooling rate) and is not negligible for some labour market institutions such as total tax rate, EPL and bargaining coordination. The section on contribution analysis (section 5.3) will more intuitively confirm that, since the importance of institutions in each country varies between the 1980s (Figure 5.4) and the 1990s (Figure 5.5).

²⁶ Technically, this vector corresponds to "pooled after-reform country trends". We have stacked the observations country by country to compose the vector. Within each block we have a trend which is 0 before any policy change and 1 after the first year following the easing of part-time regulations, 2 the second year, n the nth year, etc. In principle, we could have estimated each country block separately. However, given the relatively limited time series dimension, we stacked them and restricted the coefficients to be the same across countries in order to increase the efficiency of the estimate and to capture the *overall* effect of reforms.

²⁷ The rise in PTR induced by the regulatory change is likely to disappear in the long run, as it corresponds to the dynamics toward the new steady state implied by the reform of part-time regulation.

variable excluding the trend for the Netherlands, which experienced a much sharper increase in part-time employment than the other EU countries (Euwals and Hogerbrugge 2004)²⁸.

4. The effect of the business cycle on part-time employment in the short to medium run

Results derived from model (*a*) are presented in Table 4.1 in Annex 2. Noteworthy results include, first, that the output gap is found to significantly and negatively affect the increase in the part-time employment rate (PTR) – see column 1. When the economy operates above its potential, PTR tends to decline, or in other words, part-time employment grows to a lesser extent than full-time employment. Conversely, PTR tends to increase when the economy operates below its potential²⁹.

The negative effect of the economic cycle on PTR is also shown when running country-bycountry OLS equations, as seen in Table 4.4. The coefficient of the output gap (either contemporaneous or one-year-lagged) is negative for all countries except for Denmark, although of different magnitude across countries, and is significant at 5% for France, United Kingdom and Spain. Given the very small size of the sample for each country, the statistical inference is problematic and the value of the *t*-test might not be reliable. Therefore, we formally test the poolability of the country results: a standard Chow test does not reject the hypothesis of common output gap coefficient across countries. Panel data analysis, also called pooled-time-series estimation, increases the number of observations and degrees of freedom, allowing for better inference and providing more efficient estimators.

The negative effect of the economic cycle on PTR is consistent with the *flexibility effect* outlined in section 2.1. The counter-cyclical pattern of PTR is less marked in periods of strong economic activity than in periods of weak activity, as shown by column 7 in Table 4.1. Indeed, this would indicate that part-time employment enables employers to react flexibly in the presence of increasing economic uncertainty, e.g. in the event of recession or the very start of economic recovery. Moreover, as most part-time jobs are created in the services sector, it is not surprising that the counter-cyclical relationship is highly significant when the cycle is measured by the value-added growth in the service sector (column 6). This result is fully consistent with the aforementioned *compositional effect*.

Table 4.5 (first row) shows a highly significant and negative impact of the business cycle on the share of involuntary part-time employment. This would confirm the existence of a *flexibility*

²⁸ More technically, as the change in the regulatory framework for part-time jobs in the Netherlands occurred before the start of the sample (i.e. 1982), the "policy" variable for the Netherlands cannot really measure the effect of the *change* in regulation. It would instead simply capture the trend increase in PTR.

²⁹ Other ancillary results suggest the robustness of the findings mentioned above (see Table 4.1). The countercyclical pattern still holds when taking real GDP growth as our measure of the cycle in column 2. The cyclicality appears even stronger when using instrumental variables for the output gap to treat possible problems of endogeneity between the output gap and PTR (column 5). Column 8 suggests that the cyclical effect on part-time work is not very different between countries with high unemployment and other countries.

effect and *involuntary part-time effect*. The bulk of the effect of the cycle on involuntary PTR seems to occur within the same year. This contemporaneous effect of the business cycle holds for both the total and involuntary share of part-time work. Therefore, the expansion of economic activity recorded in the late 1990s likely reduced involuntary part-time employment, which continuously declined from 21.8% of total part-time employment in 1997 to 16.4% in 2001.

Our results generally indicate that, although strongly *significant*, the effect of the economic cycle on the part-time share appears *limited in magnitude*. The use of an annual model, for statistical reasons, may slightly underestimate the impact of the business cycle, as the intra-annual cyclical variations are not taken into account, although this measurement effect should not be overstated given that the output gap is smoothed out over a relatively short period of four quarters. A negative output gap of 1% is associated with an increase in the total PTR of around 0.06 percentage point at most. Regarding the period 1992-1999 which was characterised by continuously negative output gaps, the cycle explained less than one fifth (17%) of the total increase in PTR recorded in the European Union as a whole³⁰. This result is likely to be related to the fact that involuntary part-time work represented less than 20% of total part-time jobs in the EU in the 1990s. This also gives some support to the idea that other factors, of structural nature, which are considered in section 5, have been far more important in the explanation for the upward trend in part-time employment developments.

The remainder of this section considers the importance of the business cycle on the development of part-time work for four different groups of workers: young workers, prime age males, prime age females and older workers separately. As displayed in Table 4.3, the findings for each group are broadly confirmed by the Granger-causality approach, which is used to check the robustness of the estimates to possible endogeneity issues.

Table 4.6 presents results for model (*a*) for the part-time employment rate of young workers. The effect of the economic cycle, captured either by the output gap or real GDP movements, is found to be highly significant. The coefficient of the output gap lies between -0.15 and -0.21 across regressions. Taking a median elasticity of 0.18, the cycle would have accounted for around 25% of the total increase in youth PTR recorded in the European Union in the period 1992-1999, which is clearly stronger than the impact of the cycle on total PTR. The more acute cyclical pattern in countries displaying high unemployment might suggest that the need for higher flexibility for youth is larger in presence of strong structural rigidities. Moreover, the effect of the cycle on the share of involuntary youth part-time employment is statistically significant, but smaller compared to other groups. This would suggest that firms' labour demand generally matches the desire of young people, who are mainly looking for part-time jobs to finance their studies.

³⁰ In this period, the cycle explained 0.09 p.p. only out of the 0.51 p.p. increase in PTR recorded on average each year in the EU-15 as a whole. In the period 1992-1997, the cyclical effect reached its maximum, i.e. 23% of the total increase in PTR.

The second group we consider is prime-age males. Table 4.7 suggests that the effect of the economic cycle on the PTR for this group is again highly significant, with an elasticity of around 0.26 on the output gap (see column 3). Thus the cycle would have explained around 20% of the total increase in PTR recorded in the European Union in the period 1992-1999, which is slightly stronger than the impact of the cycle on the total PTR³¹. This is consistent with the fact that the share of prime-age male involuntary part-timers appears highly cyclical with a negative output gap of 1% increasing the proportion of involuntary part-timers by 0.34 percentage points (see Table 4.5).

For prime-age females, the effect of the cycle on the PTR is unclear and in any case very weak, as seen in Table 4.8. Although the effect of the contemporaneous output gap appears highly significant and relatively high in magnitude (column 1 and 3), it does not turn out to be significant anymore once instrumented by fixed-capital investment and total exports (column 4). When regressing the female PTR on the output gap lagged by one year so as to examine the Granger causality, the coefficient is only significant at 10% and its magnitude is almost halved (see Table 4.3). Moreover, an alternative indicator of the business cycle, such as real GDP growth, is not significant (column 2). The cycle could only explain between 8% and 15% of the total increase in female PTR recorded in the European Union in the period 1992-1999. This impact is very small and lower than the impact of the cycle on total PTR³². Thus, the upward trend in female PTR in the 1990s is likely to have mainly been the result of structural factors considered in the next section. An alternative explanation could be related to the fact that most female part-timers are working in the services sector, which is much less cyclical than industry. However, when considering specifically an indicator of cyclicality in the services sector (value-added growth), the effect on female part-time work remains insignificant.

Finally, we consider workers aged 50 and over. Table 4.9 reveals no clear effect of the cycle on the PTR for this group; neither real GDP growth nor the output gap is significant³³. This result suggests that either part-time work is particularly unattractive for this group or that employers do not typically use older workers in their attempts to increase the flexibility of their workforce through part-time work.

³¹ The effect of negative output gaps on the PTR of prime age males is found to be stronger when lagged by one year, while that of positive output gaps is strongest when contemporaneous. The negative output gap effect outweighs the positive output gap effect. One interpretation of these results again follows our flexibility effect argument: firms prefer to hire staff on flexible contracts when the economic outlook is uncertain (recession or very start of economic recovery). When the economy recovers, firms' need for part-time contracts decreases, as the uncertainty concerning future economic activity fades away gradually.

³² The relatively small impact of the output gap on involuntary female part-time work, compared to prime-age men (see Table 4.5) illustrates further the unclear effect of the business cycle on prime-age female part time employment.

³³ The only exception to this result is for relatively high unemployment countries (Spain, France, Italy and Finland) where the cycle is found to have a modest negative impact on changes in PTR.

5. The influence of institutions and other structural variables on the part-time employment rate in the longer run.

In addition to the impact of the business cycle, macro panel data analysis (model b) allows us to shed some light on the effect of the institutional and structural factors described in section 2.2 and 2.3 on part-time employment.

5.1 Using a panel of OECD countries with general labour market institutions

Covering the main OECD countries, Table 5.1 in Annex 2 shows the impact of general labour market institutions on the rate of part-time employment. As most of institutional data are only available until 1998 and display low time-series variation, we first include non-EU OECD countries into the sample in order to exploit better the cross-sectional dimension of institutions³⁴.

The findings in columns 1 to 3 do not include instrumental variables. Correcting for possible reverse causality issues (columns 4), the instrumental variable technique (2-stage GLS) broadly confirms the significance of labour market institutions in influencing PTR.

The output gap is found to be negatively correlated to PTR and appears highly significant in all equations, confirming the results displayed in section 4. The labour market participation rate is found to be positively related to PTR in equations 1, 2, 3 and 4, in line with the predictions from theory (see section 2)³⁵.

While the measure of EPL for temporary contracts is not found to be statistically significant in the explanation for the PTR, EPL for regular jobs is found to be significantly and positively related to PTR (columns 2, 3, and 4). This is consistent with very stringent EPL for regular jobs making it necessary for firms to develop flexible staffing arrangements. This is also in line with the fact that most part-time jobs are permanent.

Unionisation is found to be negatively related to PTR, possibly reflecting the reluctance of some trade unions towards the introduction of flexible work arrangements. Likewise, the unemployment benefit replacement ratio is negatively linked to PTR, signalling the possible existence of "unemployment traps", which may particularly affect part-timers given their relatively low level of earnings compared to full-timers. The labour tax rate is also negatively and highly significantly correlated with PTR. A plausible interpretation of this finding could refer to the high marginal rate of taxation borne by the second earner of a couple.

³⁴ Australia, Canada, Japan, New Zealand, Norway, Switzerland and the US.

³⁵ The surprising results found in equation 4 and 5 could be explained by the high degree of colinearity between the participation rate and some institutional variables in the context of instrumental variable estimations. This can be seen through the very high value of some coefficient.

5.2 Employing a panel of EU countries with additional institutional and structural variables

The recent literature on the effect of labour market institutions on unemployment has argued that the institutions in Europe may be different from those in other OECD countries, significantly more rigid and affect labour market output variables, such as unemployment, in significantly different ways than in the US (see Nickell et al. 2001 for the impact on overall employment and unemployment). This final subsection therefore restricts our analysis to a consideration of European countries. Furthermore, we deepen our analysis of the determinants of part-time work through the collection of other structural and institutional variables, which are expected to specifically influence part-time employment. Indeed, a further critique of the analysis in the previous specification could be, for example, that the general labour market institutions affect the labour market as a whole rather than the market for part-time work.

Table 5.2 shows the results for this focus on EU15 countries. Some findings emerge from equations 1 to 9, which are broadly confirmed when using instrumental variable techniques $(\text{column 10})^{36}$.

First, the negative influence of output gap is confirmed. Its significance is very strong in equations 1 and 9, where there are no time dummies. Indeed, time dummies, used in the other equations, tend to capture the common component of the output gap across the EU15, which is fairly important given the degree of synchronisation of European business cycles. Thus, the output gap coefficient in equations with time dummies mostly points to the country-specific component of the business cycle. As regards the magnitude of the business cycle influence, the output gap coefficient ranges from -0.07 to -0.11, which is in line with the results of section 4, albeit slightly higher.

Second, some institutions and policies appear to have a clear influence on the part-time employment rate as expected by the economic theory. The change in part-time regulation turns out to have a very significant impact on the part-time employment developments (column 4). This still holds true when excluding the Netherlands (columns 1, 2, 3, 5, 8 and 9), although the magnitude of the effect decreases somewhat. Another noteworthy result is the strong, negative and highly significant effect of child benefits on PTR, which acts as a disincentive to working part-time and makes it less attractive for individuals to (re-) enter the labour market. The reasoning behind this phenomenon may be that the benefit alters the trade-off between work and leisure towards a more family-oriented life. Broadly confirming the results of section 5.1, EPL on regular jobs is found to be highly significant, while overall EPL is only significant at a 10% level. The temporary employment rate is found to be positively correlated with the PTR (columns 7, 8 and 10), which suggests that the absence of substitution between part-time

³⁶ Results for EU15 displayed in Table 5.2 show that the impact of the output gap is negative, supporting our previous findings. It is clearly significant when the equation is estimated without time dummies (columns 1 and 9). Indeed, these dummies take out the component of the cycle common across countries.

employment and temporary employment and that these two flexibility schemes are complementary in circumventing the rigidity of European labour markets. Lastly and as in section 5.1, unemployment benefit replacement ratios are found to be negatively correlated with PTR, but the statistical significance of this variable and the magnitude of its effect are not robust across equations.

Third, some institutions do not seem to drive PTR. Unemployment benefit duration is not significant. Reflecting the theoretical uncertainty concerning its overall impact (e.g. income versus substitution effects), total labour taxes do not seem to be significant. The total labour tax rate probably remains too aggregated to identify the relative tax burden borne by part-timers compared to full-timers. In particular, it does not take into account the difference between the tax rate borne by the second earner in a household and that undergone by a single wage-earner household. Likewise, union density is not found to be significant (see column 6), which may mirror the changing attitude of trade unions toward part-time work in the EU countries in the 1990s.

Fourth, some structural factors of sociological, demographic or economic nature are also found to be an important determinant of PTR. The female participation rate is positively related to PTR. This is consistent with the fact that women seem to prefer to work part-time much more than men, in order for instance to better reconcile family and professional lives especially after their child's birth. While part-time workers are much more likely to be women, with the female part-time employment rate standing at 34% and the male rate at only 6.2% in 2001 in the EU-15, the vast majority of female part-time work is voluntary. The rate of involuntary part-time work within the EU-15 stood at 20% for men in 2001, but at a significantly lower rate of 13% for women³⁷. However, the issue of reverse causality should not be overlooked, as part-time jobs can be seen in many cases as a condition for women to enter the labour market. In order to ascertain that endogeneity problems do not lead to misspecifications and as commonly done in the literature, we first ran a two-stage least square estimation (column 10 in Table 5.2), where all the explanatory variables lagged by 2 years are used as instruments (including the female participation rate). The coefficient on the female participation rate is found to be very close to those obtained in the other equations reported in Table 5.2. Second, we followed the Grangercausality approach, where PTR is regressed by the female participation rate lagged by one year and two years. As shown in Table 5.3, the results also confirm the robustness of the previous estimates: the female participation rate lagged by one or two years remains highly significant and its coefficient is only slightly smaller than that of the contemporaneous estimate. A rise of 1 percentage point in the female participation rate would lead to an increase of around 0.3 percentage point in PTR. This is in line with the fact that female employment accounts for less than 45% of total employment in the EU15 and that almost half of the net increase in female

³⁷ This pattern of a significantly higher rate of female part-time employment and lower female rate of involuntary part-time work holds across time and for most EU-15 countries.

employment in EU15 in the late 80s and the 1990s was attributed to full-time jobs (which goes against the idea that the causality might exclusively run from female participation to part-time rate).³⁸ Finally, an alternative regression (column 5 in Table 5.2) using country trends instead of female participation also displays broadly similar results, although the multi-colinearity that country trends introduce might artificially remove the explanatory power of some institutions.

The fertility rate is also found to be strongly and positively correlated with the part-time employment rate, which suggests that part-time work creates an opportunity for women to combine taking care of their children with market work. Similarly, the schooling rate of the 15-25 aged population and the share of employment in the service sector are also found to positively affect the PTR. The relative hourly wage of part-time workers vis-à-vis full-time workers exerts a strong and highly significant effect on PTR: therefore the use of part-time work may be partly accounted for by a reduction of hourly labour costs borne by firms (see equation 11)³⁹.

Lastly, the two types of models reviewed in section 5.1 and 5.2 seem to well explain part-time developments over time and across countries. The fitted value for OECD countries is broadly close to the actual series with the exception of Denmark and the US (see Figure 5.1). The fitted value for European countries derived from Table 5.2 yields even better results, as shown by Figure 5.2^{40} .

5.3 The relative importance of the business cycle and of institutional and structural variables: a contribution analysis

The contribution analysis is one important value-added of the paper, as it measures the *relative weight* of each relevant factor over time. This justifies the macro-economic perspective and the comprehensive approach of the paper (encompassing a large set of determinants), since the contribution of relevant factors could not have been obtained with a micro analysis or with a study focusing on one specific factor of part-time employment.

The contribution analysis is carried out using the baseline version of the model (model 1 in Table 5.2). As the model is static, the contribution of each variable to the development in the part-time employment rate is the product of the regression coefficient and the change in the variable at stake over the period considered.

³⁸ Table 5.3 also suggests that the reverse causality, i.e. the impact of female participation on PTR, is higher in magnitude: in order to raise the female participation rate by 1 percentage point, a rise of 0.7 percentage point in PTR would be needed.

³⁹ The estimate of the coefficient of relative wages derived from pooled-time-series analysis might be slightly biased, as relative wage data are only available as cross-section for 1995. However, a sheer cross-sectional regression for 1995 broadly confirms the negative relationship between PTR and the relative hourly wage of part-time workers vis-à-vis full-time workers.

⁴⁰ Equation 2 in Table 5.1 and equation 2 in Table 5.2 have been chosen to make simulations because they are comparable to each other and they both use EPL indicators broken down by types of contracts (regular and temporary), which provide better results than using simply the overall EPL indicator.

Looking at Figure 5.3, the contribution of the main structural and institutional variables generally explains the development in the part-time rate in the EU countries rather well over the full period 1983-1998. This is especially true for Belgium, Germany, Spain, France and Italy, while other (unobserved) factors have also contributed to increasing part-time employment rate in the United Kingdom, Ireland, the Netherlands and Portugal. By contrast, the main structural and institutional factors do not explain the decline in the US part-time employment rate⁴¹.

As we might expect, female participation appears to be the main driving force behind the development in part-time employment. The countries experiencing the strongest increase in their part-time employment rate were those recording a surging female participation rate. However, part-time job dynamism was slightly dragged down by declining fertility, especially in Ireland, Spain and to a lower extent Italy. On the institutional side, the relaxation of the legislation on part-time jobs in Belgium, Spain and France played a major role in the rising rate of part-time work, explaining some 2 percentage points of the rise. The other institutions played a more marginal role, except in Spain and Portugal, where the reduction in EPL stringency significantly weighed upon part-time job developments, and for the Netherlands, where the cut in child benefits boosted part-time employment further.

It should be noted that the (negative) effect of the business cycle was quite negligible over the full period 1983-1998, not even reaching half a percentage point, which stresses the principally structural nature of the upward trend in part-time employment. Indeed, cyclical movements in different directions offset each other over the full period, as EU economies experienced broadly similar positions in the business cycle at the start and the end of the time sample.

In order to better identify the cyclical factors, we split the period in two sub-periods - 1983-1990 (Figure 5.4) and 1990-1998 (Figure 5.5). While the contribution of output gap was negative (between half and one percentage point) in most countries in the 1980s, it was slightly positive (of around ¹/₄ percentage point) in the 1990s, when EU economies were running slightly below their potential. Therefore, the business cycle has contributed to explaining the generally stronger developments in part-time employment in the 1990s compared with the 1980s, albeit to a modest extent.

As regards structural and institutional indicators, their individual role in the 1980s and 1990s differs across countries. However, some general features emerge. The (either positive or negative) effect of the fertility rate was stronger in the 1980s. Child benefits had an unambiguously positive contribution in 1980s, while their impact was weaker or negative in the 1990s. Conversely, the change in part-time employment laws and in EPL for regular contracts only played a real role in the 1990s.

⁴¹ This is possibly due to the flexibility of permanent contracts in the US, which makes the use of part-time jobs less necessary for firms. Even more importantly, the already high rate of female participation and the culture of much longer working hours compared with Europe could explain the stalling part-time employment rate in the US.

6. Conclusion

The negative impact of the business cycle helps to explain developments in the part-time employment rate in the short run. This paper finds that the business cycle contributed, albeit to a limited extent, to explaining the generally stronger developments in part-time employment seen in the 1990s compared with the 1980s. The countercyclical relationship between the part-time employment rate and the business cycle also suggests that part-time work may offer a means of increasing the flexibility of labour markets by allowing firms to adjust hours worked to economic conditions. Over and above this cyclical effect, the role of part-time work as a flexible form of employment is evidenced by the estimated complementarity between part-time and temporary employment, and the sensitivity of hiring part-time to the lower relative wages of these jobs.

In the longer run, the main structural variables and institutions reviewed in this paper generally explain the rise in the part-time employment rate in most EU countries over the 1983-1998 period fairly well. These factors are much less powerful in explaining part-time work developments in the United States. In absence of further evidence, it can only be hypothesized that the flexibility of permanent contracts in the US makes the use of part-time jobs less necessary for firms, while the already high rate of female participation and the culture of much longer working hours compared with Europe could explain the stalling part-time employment rate in the US.

In Europe, while female participation is found to be the main driver of the development in the part-time employment, declining fertility rates exert a dampening effect in some countries. Looking at labour market institutions, changes in legislation favourable to part-time employment are effective, having a strong impact on actual part-time employment developments. In particular, the relaxation of the legislation on part-time jobs played a key role in Belgium, Spain and France. Other institutions are found to have played some part in Spain and Portugal (easing of EPL for permanent jobs) and in the Netherlands (the cut in child benefits).

References

Barrett, G. F. and Doiron, D. J. (2001) "Working part-time: by choice or by constraint", *Canadian Journal of Economics*, Vol.34, No.4, pp.1042-1065.

Belot, M. and J. van Ours (2000) "Does the recent success of some OECD countries in lowering their unemployment rates lie in the clever design of their labour market reforms?", *CEPR Discussion Paper*. No. 2492.

Bentolila, S. and J. J. Dolado (1994) "Labour flexibility and wages: lessons from Spain", *Centro de Estudios Monetarios y Financieros: Documento de Trabajo* no 9404.

Blanchard, O. and J. Wolfers (2000) "The Role of Shocks and Institutions in the Rise of European Unemployment: the Aggregate Evidence", *Economic Journal* Vol. 78(2), pp.182-187.

Blank, R. M (1994), "The dynamics of part-time work", NBER Working Paper, No.4911.

Blundell R., Ham J, and Meghir C. (1998) "Unemployment, Discouraged Workers and Female Labour Supply", *Research in Economics*, 52, 103-131.

Buddelmeyer H., G. Mourre and M. Ward (2005a) "Part-time work in EU countries: labour market mobility, entry and exit", *ECB Working Paper*, No.460.

Buddelmeyer H., G. Mourre and M. Ward (2005b) "Recent developments in part-time work in EU countries: trends and policy", in *Labour Supply and Incentives to Work in Europe*, edited by R. Gómez-Salvador, A. Lamo, B. Petrongolo, M. Ward and E. Wasmer, Edward Elgar.

Clark K.B. and Summers L.H. (1982) "Labour Force Participation: Timing and Persistence", *Review of Economic Studies*, vol. 49, issue 5, pages 825-44.

Darby, J., Hart, R., Vecchi, M. (2001) "Labour Force Participation and Business Cycle Fluctuations: A Comparative Analysis of Europe, Japan and the United States", *Japan and the World Economy*, 13, 113-133.

Delsen, L. (1995), *Atypical employment: an international perspective. Causes, consequences and policy.* Groningen. Wolters-Noordhoff.

Delsen, L. (1998), 'When do men work part-time?'' in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Doudeijns, M. (1998), "Are benefits a disincentive to work part-time?" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Euwals, R. and M. Hogerbrugge (2004), "Explaining the growth of part-time employment: factors of supply and demand", *IZA Discussion paper* No.1124.

Farber, H. S. (1999), "Alternative and part-time employment arrangements as a response to job loss", *Journal of Labor Economics*, Vol.17, pp. 142-169.

Fagan C. and J. O'Reilly (1998), "Conceptualising part-time work - The value of an integrated comparative perspective" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Garibaldi P. and P. Mauro (2002), "Employment growth. Accounting for the facts", *Economic Policy*, April.

Genre V., R. Gomez-Salvador, N. Leiner-Killinger and G. Mourre (2003) "Non-wage components in collective bargaining" in *Wage formation in Europe*. G. Fagan, J. Morgan, F. Mongelli editors. Edward Elgar.

Ginn J. and S. Arber (1998), "How does part-time work lead to low pension income" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Giorno, C., P. Richardson, D. Roseveare and P. van den Noord (1995), "Potential output gaps and structural budget balances", *OECD Economic Studies*, N.24, 1995/I.

Guell, M and B. Petrongolo (2000), "The transition of workers from temporary to permanent employment: the Spanish case", London School of Economics CEP, DP 438.

Fisher, R. A. (1932). Statistical Methods for Research Workers, Oliver and Boyd, Edinburgh, 4th ed.

Haskel, J, Kersley, B and Martin, C (1997), "Labour Market Flexibility and Employment Adjustment: Micro Evidence from UK Establishments", *Oxford Economic Papers*, New Series, Vol.49, No.3, pp. 362-379.

Houseman, S. (2001), "Why Employers use Flexible Staffing arrangements: Evidence from an establishment Survey", *Industrial and Labor Relations Review*, Vol.55, no.1, pp. 149-169.

Jaumotte, F. (2003), "Female labour force participation: past trends and main determinants in OECD countries", *Economics Department Working Paper*, no.376, OECD, Paris.

Laroque, G and Salanié B (2003) "Participation, Fertility and Financial Incentives in France", *Paper presented at ECB/CEPR workshop* "What Explains the Pattern of Labour Supply in Europe", Frankfurt, June 2003.

Layard R., M. Barton and A. Zabalza (1980). "Married women's participation and hours", *Economica* 47(185): 51-72.

Lemaître, G., Marianna, P. and Van Bastelaer, A. (1997) "International Comparisons of Part-time Work", *OECD Economic Studies*, no. 29, pp. 139-152.

Lester, A. (1999), "Labour demand and the economic cycle" *Reserve Bank of Australia Bulletin*. February.

Long, C. (1953) "Impact of effective demand on the labour supply", *American Economic Review, Papers and Proceedings*, 43, 458-467.

Maddala G.S. and S. Wu (1999), "A Comparative Study of Unit Root Tests with Panel Data and a New Simple Test", *Oxford Bulletin of Economics and Statistics*. Special issue, 632-652.

Maloney, T. (1991). "Unobserved Variables and the Elusive Added Worker-effect.", *Economica*, 58 (230), 1736–87.

Millard S., A. Scott and M. Sensier (1999), "Business cycles and the labour market, can theory fit the facts?", Bank of England working papers.

Mincer (1962). "Labor force participation of married women: a study of labor supply" in *Aspects in Labor Economics*. H. G. Lewis editor. Princeton University Press.

Mourre G. (2006) "Did the pattern of aggregate employment growth change in the euro area in the late 1990s?", *Applied Economics*, August, 38 (15), 1783–1807.

Nickell, S, L. Nunziata, W. Ochel and G. Quintini (2003) "The Beveridge Curve, Unemployment and Wages in the OECD from the 1960s to the 1990s", in P. Aghion, R. Frydman, J. Stiglitz and M. Woodford (eds.), *Knowledge, Information and Expectations in Modern Macroeconomics: In Honor of Edmund S. Phelps*, Princeton University Press.

Nickell, S, L. Nunziata and W. Ochel (2005) "Unemployment in the OECD Since the 1960s. What Do We Know?," *Economic Journal*, Royal Economic Society, vol. 115(500), pages 1-27.

Nicoletti, G., S. Scarpetta and O. Boylaud (2000) "Summary indicators of product market regulation with an extension to employment protection legislation", *OCDE working papers* no. 266.

Nunziata, L (2005). "Institutions and Wage Determination: a Multi-country Approach," *Oxford Bulletin* of *Economics and Statistics*, University of Oxford, vol. 67(4), pages 435-466.

OECD (1999) "Focus on part-time work". Employment outlook, June.

O'Reilly, J and S. Bothfeld (2002), "What happens after working part-time? Integration, maintenance or exclusionary transitions in Britain and western Germany", *Cambridge Journal of Economics*, Vol.26, pp 409-439.

Pfau-Effinger (1998), "Culture or structure as explanations for differences in part-time work in Germany, Finland and the Netherlands" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Ruivo, M. M.P. Gonzalez and J.M. Varejao (1998), "Part-time work in Portugal and Spain" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Scarpetta, S. (1996), "Assessing the role of labour market policies and institutional settings on unemployment: A cross-country study", *OECD Economic Studies* no. 26, 43-82.

Smith, L., C. Fagan and J. Rubery (1998), "Where and why is part-time work growing in Europe?" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Sundström, M. (1991) "Part-time work in Sweden: Trends and Equity effects", *Journal of Economic issues*, March, 167-78.

Walwei, U. (1998), "Part-time work: a threat to labour standards?" in *Part-time prospects: an international comparison of part-time work in Europe, North America and the Pacific Rim.* Jacqueline O'Reilly and Colette Fagan editors. Routledge. London and New York.

Turner, D., P. Richardson and S. Rauffet (1996) Modelling the supply side of the seven major OECD economies, OECD Economics Department Working Paper No.167.

Annex 1: Data

Part-time employment

The *part-time employment rate (*PT) is defined as the percentage of part-time employment relative to total employment⁴². The annual part-time employment rate, as well as the share of involuntary part-time employment, is derived for each country from the annual European Community Labour Force Survey, which has been conducted every year in the spring since 1983 and compiled by Eurostat⁴³. Although data on part-time employment are generally available from 1983, no information is available for countries before they joined the European Union (i.e. for Spain and Portugal prior to 1986, for Austria, Sweden and Finland prior to 1995)⁴⁴. Thus, for some countries, such as Finland or Sweden, data has been back-cast using OECD Annual Labour Force Statistics, when available. These data are considered the most harmonised existing data source on part-time work. Moreover, data are available by age groups and by gender. For this study, four age/gender categories have been considered: those aged 15-24, men aged 25-49, women aged 25-49, those aged 50 and over⁴⁵.

Business cycle indicators

We mainly focus on annual output gaps stemming from the OECD Economic Outlook database. The use of the output gaps computed by the European Commission (DG-ECFIN-AMECO database) leads to very similar results⁴⁶. The output gap is measured as the percentage difference between actual GDP in constant prices, and estimated potential GDP. Its construction is precisely documented in Giorno et al. (1995) and Turner et al. (1996). The estimated potential GDP is estimated by the OECD using a production function approach for all countries but Portugal. This approach takes into account the capital stock, total factor productivity, changes in labour supply and the non-accelerating wage rate of unemployment (NAWRU) for each OECD country. By contrast, potential output for Portugal is calculated using a Hodrick-Prescott filter of actual output. It should be stressed that the estimated levels of potential output are subject to significant margins of error.

Unemployment data were also obtained from this database. Matching the frequency of part-time employment data, the business cycle is measured using annual data to match the long time series

⁴² The concept *stricto sensus* refers to employment, i.e. number of jobs rather than number of people employed. As a result, two part-time jobs held by the same individual are not counted as a full-time job but as two different jobs. Moreover, full-time and part-time employment can either correspond to self-employment or wage-earning jobs and, alternately, to fixed-term contracts or permanent positions.

⁴³ A detailed description of the sampling methods and adjustment procedures used to compile this data can be found in the latest Labour Force Survey – Methods and definitions, 1998. Part-time employment series are only available on an annual basis. Some quarterly series of part-time employment rate exist for some countries but remain rather short and display a very strong seasonal pattern, which is risky to correct given the poor time series dimension of the data.

⁴⁴ Data have been made available for 2002. The estimations in this paper used data until 2001.

⁴⁵ This split corresponds to the main age categories presented by Eurostat for the LFS.

⁴⁶ The correlation coefficient between the two measures of output gap is 0.96.

of the part-time employment rate. Although the annual frequency does not allow for a precise measurement of the dynamic adjustment to the cycle, the number of available observations appears satisfactory by pooling annual time series.

General labour market institutions

Data are taken from the recent macro-economic literature on labour markets. Data on employment protection legislation in particular for regular contracts and temporary contracts are the OECD summary series (Nicoletti et al. 2000). Data on total taxes on labour, unionisation⁴⁷ and wage bargaining co-ordination, benefit replacement rates and benefit duration are those collected by Nickell and Nunziata (2001) with an update provided by Nickell, Nunziata and Ochel (2005)⁴⁸. Most of these data on national labour market institutions are available only until 1997-1998. An indicator for the provision of child benefits (drawn from Eurostat) is also used.

Other structural variables

In addition to labour market institutions, we collected various structural data which on the basis of section 3 were likely to influence part-time employment developments. Most of these data are available for the EU countries and they are relatively harmonised across countries. Demographic variables, including fertility rates and the schooling rates of 15-24 year olds, are provided by the World Bank's World Development Indicators database. Series for total and female participation rates come from the OECD's Annual Labour Force Survey database. Data on structural aspects of national labour markets include social protection data for European countries (such as child/family benefits) which are drawn from Eurostat European Social Statistics, Expenditure and receipts dataset for 1980-1999, which gather levels of benefits in ECU/Euros for all European Countries. These variables are expressed as a percentage of GDP. Data on temporary employment as well as the share of employment in the services sector stem from Community Labour Force Surveys (Eurostat).

The wage differential between part-time and full-time employment

The wage differential (i.e. the ratio of the median hourly wage excluding overtime of part-time workers to that of full-time employees) is computed from the Eurostat Survey of the Structure of Earnings, implemented in 1995. This variable is time-invariant⁴⁹.

⁴⁷ Also called union density. This is the percentage of reported union members among wage and salaried employees.

⁴⁸ For the most recent observations, see also Nickell, Nunziata and W. Ochel (2005).

⁴⁹ The wage differential for the US used in section 6 comes from the OECD (1999).

Variables		Mean	Std. Dev.	Min	Max	Obsei	vations
Part-time variables							
Total part-time employment rates (%)	overall	13.86	8.31	3.8	38.8	Ν	216
	between		8.15	5.0	31.6	n	15
	within		2.19	3.4	21.3	Т	14.4
Part-time employment rates 15-24 (%)	overall	16.87	13.77	3.6	61.6	Ν	191
	between		14 10	5.6	41.0	n	15
	within		5 48	-7 7	37.4	Т	12.7
Part-time employment rates prime-age	overall		5.10	1.1	57.1	1	12.7
men (%)	1	2.88	2.04	0.7	9.9	Ν	181
	between		1.87	1.6	8.4	n	14
	within		0.78	-0.2	5.0	Т	12.9
Part-time employment rates prime-age	overall	26.49	16.14	5.6	68.2	N	197
women (70)	between	20.17	15.77	77	64.0	n	15
	within		2.83	19.9	37.8	Т	13.1
Part-time employment rates prime-age 50-	overall		2.05	1).)	57.0	1	15.1
64 (%)		15.55	9.27	1.4	39.6	Ν	196
	between		8.69	6.5	33.0	n	15
	within		3.27	-10.8	25.4	Т	13.1
Share of involuntary PT (%)	overall	21.38	11.53	2.3	44.8	Ν	184
	between		11.56	6.5	38.9	n	15
	within		3.91	13.1	33.4	Т	12.3
Share of involuntary PT 15-24 (%)	overall	5.37	3.21	0.2	12.6	Ν	169
	between		2.81	0.7	9.6	n	14
	within		1.51	0.4	10.0	Т	12.1
Share of involuntary PT prime-age men	overall	1 6 5 1	0.00	• •	25.0		
(%)	hetween	16.51	9.99	2.9	35.8	Ν	164
	within		9.38	5.2	31.1	n	14
	overall		3.23	4.7	24.0	Т	11.7
Share of involuntary PT prime-age women	between	12.67	7.03	1.4	29.9	Ν	173
(%)	within		7.32	4.2	27.2	n	15
	wittiili		3.04	4.7	23.4	Т	11.5
Share of involuntary PT prime-age 50-64	baturaan	3.51	1.91	0.2	8.1	Ν	167
(%)	Detween		1.88	1.3	6.4	n	14
	wittiili		0.84	1.5	6.0	Т	11.9
Economic activity	11						
Output gap (%)	overall	-1.12	2.62	-10.5	5.5	Ν	224
	between		0.82	-2.4	0.2	n	14
	within		2.50	-9.1	6.4	Т	16
GDP growth (%)	overall	2.60	2.25	-6.3	13.2	Ν	224
	between		0.85	1.6	5.1	n	14
	within		2.09	-6.1	12.9	Т	16
Institutions							
Change in part-time regulation (more	overall	0.04	2 00	0.0	17.0	NI	240
iavoulable)	between	0.74	2.00	0.0	0.5	11	240 15
	within		2.43	0.0	7.J 0 1	п	13
Change in part-time regulation (more	within		1.52	-0.0	8.4	1	10
favourable) excluding the Netherlands	overall	0.31	1.09	0.0	7.0	Ν	240
	between		0.55	0.0	1.8	n	15
	within		0.95	-1.4	5.6	Т	16
Children benefits (%)	overall	2.18	0.96	0.3	4.6	Ν	223
	between		0.98	0.4	4.0	n	15
	within		0.30	1.3	3.3	Т	14.9

 Table 3.1 Summary statistics on the variables of interest 1983-1998

-

	Variable	Mean	Std. Dev.	Min	Max	Obse	rvations
EPL	overall	2.71	1.03	0.5	4.2	Ν	224
	between		1.02	0.5	4.0	n	14
	within		0.29	2.0	3.2	Т	16
EPL for regular jobs	overall	2.60	0.94	0.7	5.0	Ν	224
	between		0.96	0.7	4.7	n	14
	within		0.17	2.0	3.0	Т	16
EPL for temporary jobs	overall	2.82	1.44	0.3	5.3	Ν	224
	between		1.38	0.3	5.0	n	14
	within		0.55	1.4	3.7	Т	16
Unemployment benefit replacement	overall	0.32	0.14	0.00	0.71	N	224
Tatios	batwaan	0.32	0.14	0.00	0.71	IN n	14
	uvithin		0.13	0.08	0.39	п	14
Unemployment benefit duration	within	0.50	0.05	0.08	1.02	I N	210
1 5	botucon	0.50	0.20	0.00	0.80	IN	210
	uvithin		0.27	0.03	0.80	п	14
Share of temporary jobs	witnin	0.10	0.08	0.23	0.71	I N	15
r i jjil	overall	0.10	0.07	0.02	0.35	IN	195
	between		0.06	0.03	0.30	n T	15
Total labour tax rate	witnin	0.20	0.02	-0.04	0.17	I N	13
	overall	0.38	0.08	0.25	0.55	IN	218
	between		0.07	0.25	0.40	n T	14
Unionisation	Within	0.45	0.02	0.31	0.47	I N	15.6
	overall	0.45	0.22	0.10	0.91	IN	206
	between		0.23	0.12	0.85	n	15
Bargaining co-ordination	Within	2.14	0.05	0.36	0.66	I N	15.8
Surganning to oraniation	overall	2.14	0.50	1.0	3.0	N	208
	between		0.46	1.0	2.9	n	13
Other structural variables	within		0.23	1.5	2.8	1	16
Female participation rate (%)		(2.10	12 50	22.0	07.7		201
	overall	62.18	13.70	33.0	8/./	N	201
	between		13.90	46.5	86.1	n	15
Total participation rate (%)	within	60.10	5.69	47.7	/6.2	I	13.4
Total participation rate (76)	overall	68.19	7.80	55.1	82.9	Ν	240
	between		7.89	57.7	80.6	n	15
Fertility rate	within		1.56	63.8	73.5	Т	16
i crunty fate	overall	1.61	0.25	1.15	2.74	Ν	240
	between		0.21	1.31	2.14	n	15
Schooling rate of 15-25 population	within		0.14	1.29	2.21	Т	16
(%)	overall	55.87	8.45	34.3	72.0	Ν	191
(70)	between		8.03	43.5	71.1	n	15
Share of employment in services	within		5.01	42.8	67.1	T	12.7
share of employment in services	overall	0.64	0.08	0.44	0.77	Ν	192
	between		0.07	0.51	0.73	n	13
Ratio part-time hourly wage / full	within		0.03	0.56	0.71	Т	14.8
time hourly wage (time invariant)	overall	0.77	0.14	0.54	1.06	Ν	240
	between		0.14	0.54	1.06	n	15
	within		0.00	0.77	0.77	Т	16

Table 3.1 Summary statistics on the variables of interest 1983-1998 (continued)

Note: unless indicated, the variables are not expressed as percentages.

ECB Working Paper Series No 872 February 2008

Annex 2: Econometric results

Table 4.1

Cyclical pattern of total part-time employment rates

Dependent variable: annual change in the part-time employment rate (p.p.) (EU-15 countries except Luxembourg⁽¹⁾ 1984-2001)

Ì	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Output gap ⁽²⁾ (GAP) Output gap (-1) (GAP-1) Output gap (-2) GDP growth GDP growth (-1) GDP growth (-2) Value-added growth in services sector Negative Output gap Positive Output gap Output gap in high unemployment countries ⁽³⁾	-0.059** (-2.10) -0.022 (-0.54) 0.04* (1.66)	-0.033 (-1.62) -0.045* (-1.90) -0.012 (-0.57)	-0.060*** (-4.51)	-0.032** (-2.31)	-0.121*** (-3.81)	-0.052*** (-2.95)	-0.064*** (-3.12) -0.055 (-1.63)	-0.053 (-3.22)*** -0.077 (-3.27)***
Instrumented variables	No	No	No	No	Yes	No	No	No
Number of observations	228	228	228	228	223	228	228	228

Sources: Eurostat, labour force surveys; OECD, Annual Labour Force Statistics and Economic Outlook (2001). ⁽¹⁾Austria, Belgium, Denmark, Germany, Greece, Finland, France, Ireland, Italy, Netherlands, Portugal and Spain, Sweden and United Kingdom. ⁽²⁾ OECD

⁽³⁾ Spain, France, Italy and Finland.

Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies to account for cross-country heterogeneity as well as a specific dummy taking into account the German reunification. When indicated, the two-stage generalised least squares (fixed-effects regression estimator with instrumental variables) have also been used to tackle possible problems of endogeneity. The instruments are the growth rate of fixed capital investments (and its value lagged by 1 and 2 years) and the growth rate of trade and services exports. The first-stage equation of the instrumental variable estimation is displayed in the Table4.2 (equation 1).

Table 4.2 First-stage equation of the instrumental variable estimation

(EU-15 countries except Luxembourg 1984-2001)

Instruments	(1)	(2)	(3)	(4)
	Dep	endent variab	le: Output gap	(%)
Growth rate of trade and services exports	0.048	0.055	0.081	
	(3.34)***	(3.53)***	(4.63)***	
Growth rate of trade and services exports lagged by 1 year		0.016	0.063	
		-0.93	(3.32)**	
Growth rate of trade and services exports lagged by 2 years		0.018	0.071	
		-1.09	(3.88)**	
Growth rate of fixed capital investments	0.078	0.075		0.083
	(6.96)***	(6.53)***		(7.44)***
Growth rate of fixed capital investments lagged by 1 year	0.087	0.084		0.082
	(8.23)***	(7.65)***		(7.68)***
Growth rate of fixed capital investments lagged by 2 years	0.071	0.071		0.071
	(7.19)***	(7.19)***		(7.03)***
Constant	-1.648	-1.835	-2.407	-1.369
	(1.80)*	(1.98)*	(4.13)***	-1.47
Observations	257	257	266	257
Number of countries	14	14	14	14

Sources: Eurostat, labour force surveys; OECD, Annual Labour Force Statistics and Economic Outlook (2001). (

Value of t-statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies.

(EO 15 countries except Euxemoourg 1901 2001)								
	Explanatory variable							
	Output gap	Output gap						
	lagged by one							
	year							
Dependent variable: Part-time employment rate								
Overall	-0.032	-0.060						
	(2.32)**	(4.51)***						
Youth	-0.078	-0.168						
	(2.21)**	(4.76)***						
Men aged 25-49	-0.028	-0.037						
	(3.28)***	(4.42)***						
Women aged 25-49	-0.058	-0.099						
	(1.95)*	(3.12)***						
Older workers	-0.002	-0.043						
	(0.08)	(1.87)*						
Dependent variable: Share	of involuntary par	rt-time						
employment								
Overall	-0.252	-0.380						
	(3.98)***	(6.58)***						
Youth	-0.050	-0.103						
	(2.85)***	(6.50)***						
Men aged 25-49	-0.268	-0.316						
	(4.69)***	(5.21)***						
Women aged 25-49	-0.090	-0.154						
	(2.43)**	(4.33)***						
Older workers	-0.022	-0.048						
	(1.60)	(3.56)***						

Table 4.3 Granger-causality from the output gap to part-time employment rates (EU-15 countries except Luxembourg 1984-2001)

Sources: Eurostat, labour force surveys; OECD, Annual Labour Force Statistics and Economic Outlook (2001).

Value of t-statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. While some heteroskedasticity has been detected, there is no evidence of serial correlation.

Table 4.4 Cyclical pattern of total part-time employment rates by country

Dependent variable: annual change in the part-time employment rate (p.p.) (OLS country-by-country equation 1984-2001)

Explanatory variable	Output	(1) gap	Output	(2) t gap lagged by one year	Number of Obs.
Austria	0.062	(0.29)	-0.156	(1.17)	6
Belgium	0.008	(0.22)	-0.041	(0.56)	18
Germany	-0.002	(0.02)	-0.038	(0.35)	18
Denmark	0.103	(0.67)	0.160	(1.16)	18
Spain	-0.084	(3.02)**	-0.012	(0.24)	14
Finland	-0.027	(1.28)	-0.008	(0.41)	18
France	-0.170	(3.96)***	-0.101	(1.53)	18
United kingdom	-0.124	(3.02)***	-0.100	(2.43)**	18
Greece	-0.084	(0.80)	0.050	(0.53)	18
Ireland	-0.041	(0.94)	-0.014	(0.20)	18
Italy	-0.075	(0.79)	-0.103	(1.13)	18
Netherlands	-0.169	(0.86)	-0.034	(0.34)	18
Portugal	-0.022	(0.40)	-0.011	(0.29)	15
Sweden	-0.096	(1.77)	-0.056	(1.02)	13
Chow's poolability test	$\gamma^{2}(13) =$	17.7(0.169)	$\gamma^{2}(13) = 11$	47 (0.572)	

Sources: Eurostat, labour force surveys; OECD, Annual Labour Force Statistics and Economic Outlook (2001).

Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%...

For Chow's poolability test, the p-value is shown in parentheses:

Note: The equations are estimated by OLS with Newey-West standard errors allowing for heteroskedastic errors and serial correlation.



Table 4.5 Cyclical pattern of involuntary part-time employment

Dependent variable: annual change in the share of involuntary part-time employment in total employment (p.p.)

	Explanatory variables					
	(1)	(2)				
Groups involved	Output gap	Output gap (-1)				
Total part-timers	-0.415*** (-7.10)	-0.265*** (-4.15)				
Youth	-0.1172*** (-7.19)	-0.0542*** (-2.99)				
Men aged 25-49	-0.337 *** (-5.43)	-0.275*** (-4.77)				
Women aged 25-49	-0.171*** (-4.70)	-0.097** (-2.58)				
Older workers	-0.0575*** (-4.12)	-0.025* (-1.82)				

(EU-15 countries except Luxembourg⁽¹⁾ 1984-2001)

Sources: Eurostat, labour force surveys. ⁽¹⁾ ⁽¹⁾Austria, Belgium, Denmark, Germany, Greece, Finland, France, Ireland, Italy, Netherlands, Portugal and Spain, Sweden and United Kingdom Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies to account for cross-country heterogeneity as well as a specific dummy taking into account the German reunification.

Table 4.6

Cyclical pattern of part-time employment rates for youth

Dependent variable: annual change in the part-time employment rate for those aged 15-24 (p.p.) (EU-15 countries except Luxembourg⁽¹⁾ 1984-2001)

(10 10 000	nuites entee	pt Editeini	coung 1	2001)		-
	(1)	(2)	(3)	(4)	(5)	(6)
Output gap (GAP) Output gap (-1) (GAP-1) Output gap (-2) GDP growth GDP growth (-1) GDP growth (-2) Negative Output gap Positive Output gap Output gap in high unemployment countries ⁽²⁾ Output gap in low and medium unemployment countries	-0.206*** (-3.09) -0.003 (-0.03) 0.075 (1.22)	-0.104** (-2.14) -0.057 (-1.19) -0.047 (-1.09)	-0.177*** (-4.92)	-0.251*** (-3.58)	-0.227*** (-3.45) -0.114 (-1.53)	-0.255*** (-3.57) -0.144*** (-3.48)
Instrumented variables	No	No	No	Yes	No	No
Number of observations	209	209	209	204	209	209

Sources: Eurostat, labour force surveys; OECD economic outlook (2001). ⁽¹⁾Austria, Belgium, Denmark, Germany, Greece, Finland, France, Ireland, Italy, Netherlands, Portugal and Spain, Sweden and United Kingdom ⁽²⁾ The high unemployment countries are Spain, France, Italy and Finland.

Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%. Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies to account for cross-country heterogeneity as well as a specific dummy taking into account the German reunification. When indicated, the two-stage least squares within estimator (fixed-effects regression estimator with instrumental variables) has also been used to tackle possible problem of endogeneity. The panel is unbalanced as part-time employment series start later for some countries (Austria, Finland and Sweden).



Table 4.7

Cyclical pattern of part-time employment rates for prime-age males

(10		es except	Luxemoou	15 17012	2001)		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Output gap (GAP) Output gap (-1) (GAP-1) Output gap (-2) GDP growth GDP growth (-1) GDP growth (-2) Negative Output gap Negative Output gap (-1) Positive Output gap (-1) Output gap in high unemployment countries (-1) Output gap in low and medium unemployment countries (-1)	-0.005 (-0.31) -0.066*** (-2.97) 0.054*** (3.67)	-0.094 (-0.79) -0.029*** (-2.64) -0.016 (-1.62)*	-0.026*** (-3.05)	-0.072*** (-4.92)	-0.020 (-1.37) -0.039* (-1.88)	-0.038*** (-2.68) -0.033* (-1.75)	-0.023 (-1.64) -0.029*** (-2.60)
Instrumented variables	No 209	No 209	No 209	Yes	No 209	No 209	No 209

Dependent variable: annual change in the part-time employment rates for prime-age males (EU-15 countries except Luxembourg⁽¹⁾ 1984-2001)

Note: For further details, see Table 4.3.

Table 4.8

Cyclical pattern of part-time employment rates for prime-age females Dependent variable: annual change in the part-time employment rate for women aged 25-49 (p.p.) (EU-15 countries except Luxembourg⁽¹⁾ 1984-2001)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Output gap (GAP) Output gap (-1) (GAP-1)	-0.127** (-2.04) 0.047 (0.52)		-0.094*** (-2.96)	-0.056 (-0.86)			
Output gap (-2)	(-0.52) -0.0162 (-0.28)						
GDP growth	(0.20)	-0.049					
GDP growth (-1)		(-1.02) -0.0199 (0.37)					
GDP growth (-2)		-0.047					
Value-added growth in services sector Negative Output gap Positive Output gap Output gap in high unemployment countries ⁽³⁾ Output gap in low and medium unemployment countries		(-0.97)			-0.049 (-0.88) -0.156** (-2.25)	-0.132*** (-2.72) -0.068 (-1.64)	-0.0473 (-1.19)
Instrumented variables	No	No	No	Yes	No	No	No
Number of observations	209	209	209	204	209	209	209

Note: For further details, see Table 4.3.

40 ECB Working Paper Series No 872 February 2008

(EU-15 cou	ntries exc	ept Luxe	mbourg	<u>/ 1984-2</u>	.001)		-
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Output gap (GAP)	-0.078 (-1.58)		-0.042* (-1.76)		-0.073		
Output gap (-1) (GAP-1)	0.008		()	0.0009	(
Output gap (-2)	0.06			(0.0.1)			
GDP growth	()	-0.055 (-1.52)					
GDP growth (-1)		-0.059					
GDP growth (-2)		-0.024 (0.73)					
Negative Output gap		(((,,,,)))				-0.001 (-0.02)	
Positive Output gap						-0.118 (-1.59)	
Output gap in high unemployment countries						()	-0.053** (-2.02)
Output gap in low and medium unemployment countries							0.008 (0.15)
Instrumented variables	No	No	No	No	Yes	No	No
Number of observations	209	209	209	209	204	209	209

 Table 4.9

 Cyclical pattern of part-time employment rates for older workers

 Dependent variable: annual change in the part-time employment rate for those aged 50-64 (p.p.)

 (DU 15 countries execut Luxembaurg⁽¹⁾ 1984 2001)

Note: For further details, see Table 4.3.

Table 5.1

OLCD pa	190	5-1996)		
	(1)	(2)	(3)	(4)
Total participation rate	0.248	0.260	0.439	0.173
	(4.09)***	(4.30)***	(8.51)***	(1.81)*
Output gap	-0.122	-0.124	-0.113	-1.578
	(-3.74)***	(-3.83)***	(-3.91)***	(-3.29)***
EPL	0.227			
	(0.71)			
EPL for temporary jobs		0.053	0.154	1.273
		(0.33)	(0.96)	(2.48)**
EPL for regular jobs		0.731	0.824	3.873
		(1.84)*	(2.05)**	(3.40)***
Unionisation	-4.764	-4.486		
	(-2.11)**	(-2.02)**		
Bargaining co-ordination	0.135	0.235		
	(0.36)	(0.63)		
Unemployment benefit replacement ratios	-4.970	-5.136		
	(-1.92)*	(-2.00)**		
Unemployment benefit duration	-0.946	-0.810		
	(-0.73)	(-0.63)		
Labour tax rate	-8.952	-9.211		
	(-2.66)***	(-2.75)***		
Constant	11.638	9.905	-12.247	6.815
	(2.12)**	(1.78)*	(-3.23)***	(0.97)
Observations	268	268	296	303
Number of countries	20	20	20	20
Instrumented	No	No	No	Yes

Effects of institutions on part-time employment rates Dependent variable: the part-time employment rate (%)

Sources: OECD; Nickell, Nunziata and Ochel (2005); Nickell and Nunziata database (2001). (1) OECD Countries included Australia, Austria, Belgium, Canada, Denmark, Germany, Finland, France, Ireland, Italy, Japan, Netherlands, New Zealand, Norway, Portugal, Spain, Sweden, Switzerland, the UK and the US.

Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies to account for cross-country heterogeneity as well as a specific dummy taking into account the German reunification. When indicated, the two-stage least squares within estimator (fixed-effects regression estimator with instrumental variables) has also been used as an alternative method to correct for possible endogeneity. The instruments used are the variables lagged by 2 years. The panel is unbalanced as part-time employment series are starting later for some countries (Austria, Finland, Norway, Sweden and Switzerland).



Table 5.2

Determinants of part-time employment rates

Dependent variable: the part-time employment rate (%)

(European countries⁽¹⁾ 1983-1998)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Output gap	-0.116	-0.075	-0.071	-0.075	-0.075	-0.0628	-0.061	-0.025	-0.090	-0.350	-0.055
	(-3.47)***	(-1.73)*	(-1.53)	(-1.93)*	(-1.46)*	(-1.50)	(-1.44)	(-0.65)	(-2.74)***	(-2.52)**	(-0.88)
Institutions	(()		((,	()		()	()	()	(
Change in part-time regulation (more				0.349							
favourable)				(6.26)***							
Change in part-time regulation (more	0.334	0.165	0.165	()	0.384	0.0506	0.115	0.125	0.375	0.029	
favourable) excluding the Netherlands	(5.68)***	(2.32)**	(2.10)**		(5.24)***	(0.51)	(1.57)	(1.94)*	(6.74)***	(0.25)	
Child benefits	-1.383	-1.679	-1.437	-1.045	-1.324	-2.146	-1.478	-1.551	-0.881	-2.999	
	(-3 56)***	(-4 53)***	(-3 49)***	(-3 10)***	(-2.49)**	(-5 14)***	(-4 09)***	(-4 52)***	(-2 42)**	(-3 88)***	
EPL	(212 2)	(0.622	()	(=,)	()	((=)	(==)	(1.00)	
			(1.73)*								
EPL for regular jobs	1.938	2.512	(11,2)	2.207	0.430	2.38	2.276	2.004	1.961	4.600	
Ç 9	(3.50)***	(5.44)***		(4.65)***	(1.05)	(5.66)***	(4.74)***	(5.02)***	(3.51)**	(3.65)***	
EPL for temporary jobs	-0.059	0.072		-0.126	-0.390	0.0306	0.050	0.171	-0.337	-0.684	
* 55	(-0.35)	(0.42)		(-0.93)	(-1.81)*	(1.73)*	(0.30)	(1.12)	(-2.25)**	(-1.51)	
Unemployment benefit replacement	-0.789	-1 877	-1 448	-1 029	-15.17	-3 77	-2.058	-1 636	-2.622	-3 631	
ratios	(-0.37)	(-2.09)**	(-1.39)	(-1.25)	(-3 29)***	(-2.89)**	(-2.36)**	(-2,13)**	(-2.62)***	(-1.63)	
Unemployment benefit duration	-0.253	-0.790	-1.090	0.099	1 736	-1.61	-0.853	-1 431	-0.087	5 340	
1 5	(-023)	(-0.80)	(-0.99)	(0.12)	(1.18)	(-1.53)	(-0.89)	(-1.69)*	(-0.09)	(1.67)*	
Share of temporary jobs	(025)	(0.00)	(0.57)	(0.12)	(1.10)	(1.55)	6 514	7 285	4 168	15 593	
F S S							(1.99)**	(2.52)**	(1.19)	(2.17)**	
Total labour tax rate						8 224	(1.))	(2.02)	()	(2.17)	
						(1.61)					
Unionisation						-1 728					
						(-0.56)					
Other structural variables						(0.00)					
Female participation rate	0.374	0.401	0.381	0 354		0 3497	0 397	0.317	0.271	0 327	0.325
· · · · · · · · · · · · · · · · · · ·	(16.5)***	(12 14)***	(10.30)***	(12 25)***		(10.22)***	(12 40)***	(0 / 3)***	(7.84)***	(5.18)***	(12.06)***
Fertility rate	3 425	4 424	(10.50)	2 313	1 296	(10.22)	(12.10)	8 230	(7.04)	6.489	(12.00)
	(4.05)***	(5 /8)***	(5 35)***	(3 20)***	(3.88)***	(5 30)***	(6.05)***	(6.96)***		(2.86)***	
Schooling rate of 15-25 population	0.029	0.079	0.033	0.082	0.028	0.0568	0.072	0.051	0.010	0.169	0.076
5	(2 63)***	(3 /1)***	(1.34)	(3 70)***	(1.04)	(2 10)**	(3 22)***	(2 32)**	(0.45)	(3 11)***	(2 27)**
Share of employment in services	(2.05)	(3.41)	(1.54)	(3.77)	(1.04)	(2.1))	(3.22)	22.52)	7 767	30 200	(2.27)
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~								(2.85)***	(1.18)	(1.67)*	
Ratio part-time hourly wage / full-								(2.05)	(1.10)	(1.07)	-11.850
time hourly wage (time invariant)											(1 04)***
Constant	-9.77	-14 382	-13 637	-11 033	18 16	-17 557	-15 275	-12 834	-2.079	-13 724	-4 409
Constant	(-3.31)	(3.24)**	(2 45)**	(2 75)***	(2 70)***	-17.337 (_2.48)**	-13.275	(7.31)***	-2.079	(4.04)***	(1.43)
Observations	172	172	172	172	172	154	170	154	154	154	231
Number of countries	14	1/2	1/2	14	1/2	14	14	14	14	14	15(2)
Time dummies	no	Ves	Ves	Ves	1 <del></del>	Ves	Ves	Ves	1 <del></del>	Ves	Ves
Country trends	no	no	yes	yes	Vec	yc5	po	po	no	no	no
Instrumented variables	no	no	no	no	no	no	no	no	no	ves	no

Sources: Eurostat Labour force surveys; Eurostat European Social Statistics; OECD economic outlook (2001); OECD labour market database; World Bank's World Development Indicators database; Nickell, Nunziata and Ochel (2005); Nickell and Nunziata database (2001).

⁽¹⁾ Countries included all the European Union member states except Luxembourg (Belgium, Denmark, Germany, Greece, Spain, France, Ireland, Italy, Netherlands, Austria, Portugal, Finland, Sweden and the UK).

⁽²⁾ The US is also included to increase the cross-sectional dimension, as the ratio of part-time hourly wage to full-time hourly wage is time invariant, corresponding to its 1995 value (Eurostat survey of structure of earnings). For the same reason, equation 11 was estimated without country dummies (GLS with time dummies).

Value of t- statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally estimated by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation. Each equation contains country dummies, intercepts and a specific dummy taking into account the German reunification. When indicated, the two-stage least squares within estimator (fixed-effects regression estimator with instrumental variables) has also been used as an alternative method to correct for possible endogeneity. The instruments used are the variables lagged by 2 years. The panel is unbalanced as data are starting later for some countries (Austria, Finland, Norway and Sweden).

#### Table 5.3

(E	0-15 countries except Lu	xembourg 1984-2001	)		
		Explanatory variable			
	contemporaneous	lagged by 1 year	lagged by 2 years		
Dependent variable: Total pa	art-time employment rate				
Female participation	0.358	0.318	0.314		
	(17.22)***	(15.71)***	(15.43)***		
Observations	223	210	197		
Number of countries	14	14	14		
Dependent variable: Female	participation rate				
Total part-time rate	1.388	1.382	1.269		
	(20.73)***	(20.30)***	(15.74)***		
Observations	223	211	199		
Number of countries	14	14	14		

#### **Granger-causality from the female participation rate to part-time employment rate** (EU-15 countries except Luxembourg 1984-2001)

Sources: Eurostat, labour force surveys; OECD, Annual Labour Force Statistics (2001).

Value of t-statistics in parentheses: * significant at 10%; ** significant at 5%; *** significant at 1%.

Note: The equations are generally by feasible generalised least squares allowing for heteroskedastic errors and common-across-group first order serial correlation.

# Figure 5.1: The fitted value of part-time employment rate for OECD countries (model 2 of Table 5.1) (%)







Figure 5.3: Change in part-time employment rate (in black) and contribution of each determinant over the full-period (1983-1998) (model 1 of Table 5.2) (percentage points)







Figure 5.5: Change in part-time employment rate and contribution of each determinant in the 1990s period (1990-1998) (model 1 of Table 5.2) (percentage points)



#### **European Central Bank Working Paper Series**

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.europa.eu).

- 827 "How is real convergence driving nominal convergence in the new EU Member States?" by S. M. Lein-Rupprecht, M. A. León-Ledesma, and C. Nerlich, November 2007.
- 828 "Potential output growth in several industrialised countries: a comparison" by C. Cahn and A. Saint-Guilhem, November 2007.
- 829 "Modelling inflation in China: a regional perspective" by A. Mehrotra, T. Peltonen and A. Santos Rivera, November 2007.
- 830 "The term structure of euro area break-even inflation rates: the impact of seasonality" by J. Ejsing, J. A. García and T. Werner, November 2007.
- 831 "Hierarchical Markov normal mixture models with applications to financial asset returns" by J. Geweke and G. Amisano, November 2007.
- 832 "The yield curve and macroeconomic dynamics" by P. Hördahl, O. Tristani and D. Vestin, November 2007.
- 833 "Explaining and forecasting euro area exports: which competitiveness indicator performs best?" by M. Ca' Zorzi and B. Schnatz, November 2007.
- 834 "International frictions and optimal monetary policy cooperation: analytical solutions" by M. Darracq Pariès, November 2007.
- "435 "US shocks and global exchange rate configurations" by M. Fratzscher, November 2007.
- 836 "Reporting biases and survey results: evidence from European professional forecasters" by J. A. García and A. Manzanares, December 2007.
- 837 "Monetary policy and core inflation" by M. Lenza, December 2007.
- 838 "Securitisation and the bank lending channel" by Y. Altunbas, L. Gambacorta and D. Marqués, December 2007.
- 839 "Are there oil currencies? The real exchange rate of oil exporting countries" by M. M. Habib and M. Manolova Kalamova, December 2007.
- 840 "Downward wage rigidity for different workers and firms: an evaluation for Belgium using the IWFP procedure" by P. Du Caju, C. Fuss and L. Wintr, December 2007.
- 841 "Should we take inside money seriously?" by L. Stracca, December 2007.
- 842 "Saving behaviour and global imbalances: the role of emerging market economies" by G. Ferrucci and C. Miralles, December 2007.
- 843 "Fiscal forecasting: lessons from the literature and challenges" by T. Leal, J. J. Pérez, M. Tujula and J.-P. Vidal, December 2007.
- 844 "Business cycle synchronization and insurance mechanisms in the EU" by A. Afonso and D. Furceri, December 2007.
- 845 "Run-prone banking and asset markets" by M. Hoerova, December 2007.

- 846 "Information combination and forecast (st)ability. Evidence from vintages of time-series data" by C. Altavilla and M. Ciccarelli, December 2007.
- 847 "Deeper, wider and more competitive? Monetary integration, Eastern enlargement and competitiveness in the European Union" by G. Ottaviano, D. Taglioni and F. di Mauro, December 2007.
- 848 "Economic growth and budgetary components: a panel assessment for the EU" by A. Afonso and J. González Alegre, January 2008.
- 849 "Government size, composition, volatility and economic growth" by A. Afonso and D. Furceri, January 2008.
- 850 "Statistical tests and estimators of the rank of a matrix and their applications in econometric modelling" by G. Camba-Méndez and G. Kapetanios, January 2008.
- 851 "Investigating inflation persistence across monetary regimes" by L. Benati, January 2008.
- 852 "Determinants of economic growth: will data tell?" by A. Ciccone and M. Jarocinski, January 2008.
- 853 "The cyclical behavior of equilibrium unemployment and vacancies revisited" by M. Hagedorn and I. Manovskii, January 2008.
- 854 "How do firms adjust their wage bill in Belgium? A decomposition along the intensive and extensive margins" by C. Fuss, January 2008.
- 855 "Assessing the factors behind oil price changes" by S. Dées, A. Gasteuil, R. K. Kaufmann and M. Mann, January 2008.
- 856 "Markups in the euro area and the US over the period 1981-2004: a comparison of 50 sectors" by R. Christopoulou and P. Vermeulen, January 2008.
- 857 "Housing and equity wealth effects of Italian households" by C. Grant and T. Peltonen, January 2008.
- 858 "International transmission and monetary policy cooperation" by G. Coenen, G. Lombardo, F. Smets and R. Straub, January 2008.
- 859 "Assessing the compensation for volatility risk implicit in interest rate derivatives" by F. Fornari, January 2008.
- 860 "Oil shocks and endogenous markups: results from an estimated euro area DSGE model" by M. Sánchez, January 2008.
- 861 "Income distribution determinants and public spending efficiency" by A. Afonso, L. Schuknecht and V. Tanzi, January 2008.
- 862 "Stock market volatility and learning" by K. Adam, A. Marcet and J. P. Nicolini, February 2008.
- 863 "Population ageing and public pension reforms in a small open economy" by C. Nickel, P. Rother and A. Theophilopoulou, February 2008.
- 864 "Macroeconomic rates of return of public and private investment: crowding-in and crowding-out effects" by A. Afonso and M. St. Aubyn, February 2008.
- 865 "Explaining the Great Moderation: it is not the shocks" by D. Giannone, M. Lenza and L. Reichlin, February 2008.
- 866 "VAR analysis and the Great Moderation" by L. Benati and P. Surico, February 2008.

- 867 "Do monetary indicators lead euro area inflation?" by B. Hofmann, February 2008.
- 868 "Purdah: on the rationale for central bank silence around policy meetings" by M. Ehrmann and M. Fratzscher, February 2008.
- 869 "The reserve fulfilment path of euro area commercial banks: empirical testing using panel data" by N. Cassola, February 2008.
- 870 "Risk management in action: robust monetary policy rules under structured uncertainty" by P. Levine, P. McAdam, J. Pearlman and R. Pierse, February 2008.
- 871 "The impact of capital flows on domestic investment in transition economies" by E. Mileva, February 2008.
- 872 "Why do Europeans work part-time? A cross-country panel analysis" by H. Buddelmeyer, G. Mourre and M. Ward, February 2008.



